Copyright © 2023, Emerald Publishing Limited This is the version of the article accepted for publication in *Journal of Accounting in Emerging Economies* published by Emerald. https:// doi.org/10.1108/JAEE-09-2020-0247

This AAM is provided for your own personal use only. It may not be used for resale, reprinting, systematic distribution, emailing, or for any other commercial purpose without the permission of the publisher

This version downloaded from SOAS Research Online: http://eprints.soas.ac.uk/39511

Internal Governance Mechanisms and Information Value of Banks Earnings

Marwa Elnahass Newcastle University Business School, Newcastle University, UK. Muhammad Tahir SOAS University of London, UK Noora Yusuf College of Business Administration, University of Bahrain, Bahrain. Aly Salama Northumbria University Newcastle, UK.

Abstract

Purpose

This study examines the association between internal corporate governance mechanisms (i.e. board of directors and audit committee) and the information value of bank earnings. We comparatively assess this association across different bank types, Islamic versus conventional banks. We also investigate the mediating effect of Shariah governance.

Design/methodology/approach

We utilise a unique and an international sample of 723 bank-year observations representing 100 listed banks from 16 countries during the period of 2007 to 2015. We investigate the characteristics of the board of directors and audit committee (i.e. size and independence) and employ three core analyses for earnings informativeness (i.e. earnings persistence, cash flow predictability and reliability of loan loss provisions). Additional analyses address Shariah supervisory boards' size, financial expertise, and multiple outside directorships. We use the random-effect GLS estimation technique and provide several robustness checks and sensitivities.

Findings

We find that, on average, having large and independent boards (and audit committees) increases the informativeness of reported earnings for banks. Conditional on bank type, our results report strong evidence for differential effects across the two alternative banking systems. In Islamic banks, large and independent board of directors (and audit committees) are positively associated with all measures of information value. There is insignificant evidence for conventional banks. However, Shari'ah supervisory boards show no significant effect on the reported earnings' informativeness.

Originality/value

This is the first study, to the best of our knowledge, that empirically and comparatively assesses the information value of reported earnings in association with effective internal governance while recognizing the institutional characteristics of different bank types. We offer new insights to policy makers, investors and other stakeholders located within countries operating on a dual banking system. The results could help regulators to improve their rules/guidance related to double-layer governance and financial reporting quality.

Keywords: Corporate governance; Information Value; Bank type, Financial reporting

JEL Classifications: C23, G01, G21, G28, L50, M4

1. Introduction

The quality of corporate financial reporting and systems of governance have been the subject of extensive research and debates representing key factors influencing firm's stability and value. The number of accounting scandals that the twenty-first century has witnessed, such as Enron and WorldCom, has caused severe damage to investors' confidence in financial reporting systems, and raised many concerns regarding managerial opportunism. The governance of banking institutions came under stricter scrutiny after the onset of the financial crisis of 2007-2009, because bank poor governance has been listed among major causes of the financial turmoil and loss of public trust (Adams and Mehran, 2012; Li et al. 2022). Although banks contribute to the strength of the national economy, some banking practices showed either inadequate quality or a lack of transparency in published annual reports which tend to promote discretionary practices, such as aggressive earnings management and fraudulent reporting (e.g. the cases of Lehman Brothers, and Bear Stearns Company).

Corporate governance in the banking industry is different from that in non-financial sectors due to its underlying principles. The banking sector is highly regulated and more complex than other areas of business because the external regulators of banks act on behalf of stakeholders such as depositors, investors, and creditors, ensuring that the banks are sound and work in their interests. The complexity and diversity of banking financial instruments and transactions lead to substantial information asymmetries (De Andres and Vallelado, 2008). The importance of high levels of regulation in the banking sector lies in the fact that ineffectual financial reporting can lead to substantial opportunism and banking instability (Adams and Mehran, 2003; Elnahass et al. 2018).

Boards of directors alongside audit committees are marked as key *"internal governance mechanisms"* in banks which play a vital role in making decisions that balance the requirements for financial stability and quality of reported earnings by banks through optimal risk management and effective monitoring (Alharbi et al. 2022). The board of directors' efficiency in the banking industry is different than in other sectors due to the highly regulated and complex system in the sector, which requires high monitoring and counselling (Trinh et al. 2020). Board members work on maximizing the wealth of shareholders through decision making and monitoring the managers, especially when their activities conflict with the shareholders' interests. This latter point concerns a notable phenomenon called the agency issue, which arises from a conflict in interests between managers and shareholders, but which may be reduced by monitoring on the part of the board of directors.

Much emphasis has also been placed on the audit committee's role in overseeing the financial reporting process and preventing fraudulent accounting statements. A board of directors delegates to the audit committee the duty of monitoring and controlling the financial reporting process, which involves oversight of financial reporting, internal control, and external audit activity (Joshi and Wakil, 2004; Pomeroy and Thornton 2008; Beasley et al. 2009). Both, the board of directors and audit committees can use their wider experience to promote the best interest of both the company and the shareholders and a strong representation of non-executive directors is able to monitor managers' action more effectively, consequently restricting their self-interest (Peasnell et al. 2005, Chiu et al. 2012 and Tahir et al. 2019). According to Watts and Zimmerman (1978) and Bushman and

Smith (2001), good financial reporting limits managerial self-interest behaviour and aligns managers' interests with those of shareholders. To make accurate decisions, users of financial reports need high-quality information which must be reliable and trustworthy.

Several studies in the literature have underlined the critical role of good corporate governance to limit manipulative earnings management practices and protect shareholders' interests (e.g. Xie et al. 2003; Iatridis and Kadorinis 2009; Salem et al. 2021). The literature on banking governance commonly shows how critical the role of the board of directors and audit committee are in mitigating bank risk, and driving performance, earnings management, and stock market valuation (Pathan and Faff, 2013; Berger et al., 2014; Dong et al., 2017; Owen and Temesvary, 2018; Elnahass et al., 2020; Trinh et al., 2020; Elnahass et al., 2022a). However, prior literature still lacks evidence on the role of internal governance on the informativeness of reported earnings within the banking sector. Academic and regulatory pressure is exerted to promote the informativeness of bank earnings, helping to mitigate earnings management practices and to promote long-term banking resilience (Srivastav and Hagendorff, 2016).

Moreover, there is a lack of empirical studies in the literature examining the role of bank type (i.e., Islamic versus conventional banks) on corporate governance- earnings informativeness nexus.¹ Accordingly, studying the effect of board members and audit committee, as key internal governance mechanisms, on banks' informative value of earnings while considering the effect of bank type became an important matter in the corporate governance literature (Pathan and Faff, 2013; Mollah et al., 2017; Trinh et al., 2020). This effect could be pervasive for specific banking systems like Islamic banks, marked by religious business practices, high agency costs, and complex internal governance and the quality of information for reported earnings particularly within religious establishments (e.g., Islamic banking) has become prolific.

This study covers the existing gaps in prior literature through employing empirical assessments of whether enhanced internal corporate governance can increase the value of published bank earnings information by examining board of directors' and audit committees' size and independence. We take a step further to examine this predicted association across Islamic versus conventional banks. Islamic banks are distinguished from conventional banks by several characteristics in their business models. The operations of the Islamic banking industry are principally driven by a constrained business model and complex agency costs, which inherits both religious orientations and moral accountability values alongside legal responsibilities (Abdelsalam et al. 2016, 2021). Furthermore, governance in Islamic banking is more extensive than that in conventional banking. Besides the traditional governance mechanisms used in conventional banks (i.e., board of directors and audit committees), Islamic banks operate on an additional (i.e., non-traditional) governance structure, with the existence of the Shariah supervisory board (SSB).² This internal (extra) layer of governance represents scholars

¹ Islamic banks are those banks that follow Islamic Shariah principles in their business transactions. These banks operate on a banking model which prohibits usury, excessive uncertainty and speculation while encourages risk and profit-sharing between the bank and its depositors. Conventional banks refer to traditional commercial banks which operate on an interest basis (Elnahass et al. 2018).

² The Shariah supervisory board (SSB) acts as a monitoring mechanism to carry out an independent audit and issue a separate report as part of the bank's financial statements. SSB members may also have to review additional information and reports, such as operating

in Islamic legitimacy who monitor the banks' activities and funding decisions.

Our sample comprises 723 bank-year observations from 100 listed banks (61 conventional and 39 Islamic) in 16 countries operating on a dual banking system for the period 2007 to 2015. The empirical setting of this study uses several alternative models to measure the information value of earnings such as earnings persistence, predictability of future cash flows and the reliability of loan loss provisions. We examine the size and independence of both the board and audit committees for the whole sample in addition to the two bank types. Given the constrained business model and the additional governance layer adopted by Islamic banks, our premise is that the quality of information for reported earnings is likely to be higher in Islamic banks than in conventional banks. This prediction is also in line with prior evidence (Elnahass et al. 2014, 2018; Abdelsalam et al. 2016, 2020; Lassoued et al. 2018; Salem et al. 2021), showing that Islamic banks engage in a higher financial reporting quality and lower managerial opportunism as compared to their conventional counterparts. That premise is also consistent with prior literature, which documents that religious orientation and robust institutional environments ultimately shape corporate behaviour and mitigate aggressive earnings management (McGuire et al. 2012; Kanagaretnam et al. 2015; Elnahass et al. 2022a). Moreover, Islamic banks commonly act under a predominant set of social norms.³

Our results show strong evidence that internal governance mechanisms play a catalytic role in promoting high quality and informative reported earnings for the whole sample of banks (i.e. irrespective of the bank types). We find that, on average, having both large and independent boards of directors (and audit committees) can significantly increase earnings persistence, cash flow predictability, and reliability of loan loss provisions. Conditional on the bank type, our results show significant differences between Islamic and conventional banks. Consistent with our predictions, we find that positive association between large as well as independent board of directors (and audit committees) and the information quality of earnings is more pronounced for Islamic banks when compared to their conventional counterparts. In particular, unlike conventional banks, large and independent board and audit committee of Islamic banks can significantly enhance the persistence of earnings, the ability of current earnings to predict future cash flows and the reliability of loan loss provisions. Additional analysis for the effect of Shariah governance (i.e. Shariah supervisor boards' size, financial expertise and multiple outside directorship) indicate an insignificant impact on increasing the information value of earnings for Islamic banks. Our main findings are robust in several model sensitivities and specifications (i.e., propensity score matching and GMM).

Our findings contribute to the broad strands of literature on earnings quality and corporate governance. First, to our knowledge, this is the first empirical study to identify the combined effects of internal governance mechanisms on information quality of earnings in the banking sector. We provide international evidence while

and financial reports and policies (Abdelsalam et al. 2020). This board is appointed during the annual general assembly, and its members are likely to be recommended by the board of directors and approved by the shareholders.

³ The social norm theory provides detailed grounds for shaping individual economic attitudes (Akerlof 1980; Kohlberg 1984). Social norms represent the prevailing code of conduct and ethics jointly shared by a group of individuals. This code drives forces and mechanisms for individuals. Compliance with norms and group expectations is subject to community support and acceptance, while non-compliance would promote social discrimination.

recognizing the effect of bank type alongside the mediating role of Shariah governance. This study goes beyond the evidence presented by Leventis and Dimitropoulos, 2012; Elamer et al., 2019), through offering new insights on the impact of internal governance from an earnings informational context (i.e. earnings persistence, predictability and reliability). Second, none of the prior literature has investigated the above predicted association among the two bank types. Previous studies have examined the different governance-related characteristics and earnings management among the two bank types without considering information value of earnings (Abdelsalam et al. 2016; Elnahass et al. 2022a). As such, our research is among early attempts in the comparative literature of Islamic versus conventional banking. Finally, previous studies have mainly considered the role of additional monitoring (i.e. Shariah governance) within the context of earnings management (e.g. Elnahass et al. 2018, 2022a). Other studies examined the individual effect of Shariah supervisor board on bank risk-taking, performance and social reporting (e.g., Abdul Rahman and Bukair, 2013; Almutairi and Quttainah, 2017). Together, the studies have failed to consider the mediating effect of such extra governance mechanism on the information quality of reported earnings. As such, we extend prior evidence by examining the role of Shariah governance.

This study provides important implications for policymakers and various sets of stakeholders engaging with global banking sectors. We highlight the importance of recruiting large and independent boards as well as audit committees to improve the information quality for reported earnings based on an international context. Through studying the effect of the bank type, this study highlights the impact of institutional characteristics and alternative business models in promoting the informativeness and quality of reported earnings across dual banking systems. The significant variations in results among two bank types offer new insights to the existing evidence in literature (e.g. Abdelsalam et al. 2016, Elnahass et al. 2014, 2018, 2022a) that internal mechanisms are among important determinants of the higher financial reporting quality for Islamic banking as compared to conventional banking. Hence, our findings provide strong evidence that could raise the attention of regulators and investors regarding the role of effective governance and institutional characteristics on the information quality of bank earnings.

The remainder of the paper is organized as follows; section 2 provides the background and hypotheses development. Section 3 describes the research methodology. Section 4 presents the empirical results and robustness tests. Finally, section 5 concludes.

2. Background and hypotheses development

Following the collapse of several prominent financial institutions, such as Lehman Brothers, Bear Stearns and Company and Bank of Credit and Commerce International, there has been a growing concern that corporate governance mechanisms were weak and ineffective, resulting in poor board oversight and detrimental bank value. In line with the agency theory, self-interest and external rewards motivate managers to behave opportunistically. Agency conflicts may arise from the separation of ownership and control (Jensen and Meckling 1976; Fama and Jensen 1983). The problem of "information asymmetry", where the agents have more access to the company's information than the principals (Arnold and Lange 2004), creates more earnings management opportunities and reduces the information quality for earnings. Information asymmetry

complicates the agency conflicts, as managers can manipulate the information they disclose, and owners might not monitor and evaluate managers' actions accurately. Poor financial reporting quality and the underpinning earnings management can be viewed as core agency costs (Jiraporn et al. 2008). In line with the agency theory, managerial self-serving and opportunistic behaviour can be limited by establishing formal and effective corporate governance mechanisms. The agency theory considers corporate governance mechanism(s) as one of the classical cures in controlling conflict of interests between agents and principals (Shleifer and Vishny 1997; Brennan 2006). Effective monitoring by board and audit committee are indications of a sound internal governance system, which can help to reduce agency costs through greater monitoring activities (Ronen and Yaari 2008; Kent et al. 2010; Talavera et al. 2018; Trinh et al. 2020).

Company reports, providing information about a company's financial position and performance, serve as the primary source of accounting information for stakeholders. Such knowledge is highly influential for decision-making by investors, creditors, regulators, and other stakeholders (Beyer et al., 2010; Chen et al., 2018). The importance of accounting information arises from its dual role; informativeness to all parties and the role of stewardship (Feltham et al., 2006; Cascino et al., 2014). High quality information allows users to make informed decisions based on the perceived risks and future cash flows. Accurate accounting information is crucial for well-functioning markets, as it reduces information asymmetry between managers and capital providers (Biddle and Hilary, 2006), minimises the risks of moral hazards and adverse selection (Li, 2008), and enhances the efficiency of capital allocation (Bhattacharya et al., 2003).

Prior studies investigating the information value of earnings have examined it in relation to improvements in internal control (Altamuro and Beatty, 2010), legal, extralegal, and political institutional factors (Kanagaretnam et al., 2014), corporate social responsibility practices (García-Sánchez and García-Meca, 2017), and regulatory enforcement actions (Delis et al., 2018). Within the context of corporate governance, some studies examine the role of individual traditional governance mechanisms in earnings management, such as board size (Beasley 1996); board independence (Bédard et al. 2004; Dimitropoulos and Asteriou 2010); audit committee financial expertise and diversity (Zalata et al. 2018). Other previous studies investigate the association between governance index and earnings management (Iatridis and Kadorinis 2009; Leventis and Dimitropoulos 2012). Therefore, detailed examination of the information value of earnings in relation to internal governance mechanisms is still lacking. Such gaps in the literature are particularly evidential for the banking industry.

Banks are generally characterized by a high degree of opaqueness in transactions and activities. As a result, corporate governance in the banking sector exhibits its own attributes and characteristics. This is because of the unique features of the banking system, which aggravate governance problems and can lessen the effectiveness of traditional governance structures like boards and audit committees (Levine, 2004; Laeven, 2013; Alharbi et al. 2022), relative to non-financial firms. Because depositors are the major capital providers, their interests may deviate from shareholders' interests. The high leverage provides incentives for managers to invest in risky projects, as debtholders bear more of the excessive risk and not shareholders (Elnahass et al. 2022b). Banks are also exposed to a higher degree of different types of risks (e.g. credit, liquidity, insolvency, operational and, asset risks), when compared to non-financial institutions (John et al., 2016; Trinh et al, 2020). Another unique

bank characteristic that contributes to the importance of governance in banks is the high information asymmetry which arises from managers ability to hide valuable information about loan quality and structure, making it difficult to monitor their activities and this in turn threatens the traditional corporate governance mechanisms (Levine, 2004; Abdelsalam et al. 2021).

These unique bank features demonstrate the need for more effective and distinct corporate governance mechanisms for banks in order to control the high agency costs. Confirming this view, the Basel Committee on Banking Supervision (BCBS) states that the banking industry has its own rules concerning corporate governance involving "the manner in which the business and affairs of banks are governed by their boards of directors and senior management" (Basel Committee on Banking Supervision, 2006, p. 4). This implies that due to the special nature of banking institutions, internal governance mechanism plays a catalytic role in the way they conduct their business and meet their obligations.

2.1. Board of directors and information value of earnings

According to agency theory, the board of directors is considered as a primary governance mechanism employed to address conflicts of interests arising from relationships between agents and principals. Beasley (1996) claims that the board of directors is responsible for ensuring that high-quality financial information is available to all stakeholders. Prior studies have provided evidence that effective monitoring by board of directors enhances the quality of financial reporting (Leventis and Dimitropoulos, 2012; Quttainah et al., 2013; Abdelsalam et al., 2016; Elnahass et al., 2018). However, these studies have assessed the quality of financial reporting mainly from an opportunistic earnings management perspective and hence, the impact of the board of directors' characteristics on the information value of earnings has not been broadly investigated.

From an information perspective, the quality of financial reporting increases as managers discloses more accurate information and ensure that the financial information precisely reflect the firms' future earnings and cash flows. This informativeness role of accounting information has been highlighted by the conceptual framework for financial reporting developed by the International Accounting Standards Board (IASB) where it emphasizes that the objective of financial reporting is to provide useful information about the reporting entity to existing and potential investors, lenders, and other creditors. These users rely on financial statements to help them assess the risks and prospects for future cash flow to the entity (International Financial Reporting Standards, 2018). Given that the board of directors is responsible to ensure that high-quality accounting information is available to all users, effective governance through large and independent board members would be expected to enhance the information value of reported earnings.

The composition of the board of directors may also be a factor in strengthening the value of earnings information. Board size has a bearing on the prevalence of opportunistic behavior (Elnahass et al., 2022a; Tahir et al. 2019). A large board may encourage the putting forward of alternative views which otherwise would not have been considered. According to the resource dependence theory, a larger board is "a provider of resources, such as legitimacy, advice and council links to other organizations, etc." (Hillman and Dalziel 2003, p. 383) and

therefore enhances the skills, expertise, and knowledge needed to exert effective monitoring over earnings management practices (Xie et al. 2003; Peasnell et al. 2005).

Besides board size, the board's independence is essential (Filatotchev and Wright 2011). According to the agency theory, independent directors can exercise sovereign judgement to protect shareholders' interests when an agency conflict is present (Jensen and Meckling 1976). Given the need to develop and maintain a reputation in the labour market, and since independent directors bring valuable expertise and potential networks that could benefit the firm (Fama and Jensen 1983; Pathan and Skully 2010), boards dominated by independent directors are better positioned to monitor and control managers' activities (Fama and Jensen, 1983). A larger board of directors with a high proportion of independents is expected to enhance the information value of bank earnings (Xie et al., 2003; Davidson et al., 2005; Kang and Kim, 2012; González and García-Meca, 2014) and can bring enhanced skills, knowledge, and expertise to exert effective monitoring over financial reporting. If independent directors on the board enhance monitoring, they should also be associated with lower use of earnings management (Cornett et al. 2009). Beasley (1996), for example, examines whether including larger proportions of outside members on the board reduces the likelihood of financial statement fraud and finds that non-fraud firms have boards with significantly higher percentages of outside members than those of fraud firms. Dechow et al. (2010) find that firms manipulating earnings are more likely to have boards with a lower proportion of independent members.

We conjecture that large board size and more independent boards are likely to monitor/mitigate managerial opportunism and promote high quality information for reported earnings. However, as we extend our insights to the effect of the bank type, we also conjecture that for Islamic banks operating on a complex and constrained banking model, the role of the board of directors in controlling agency problems should be more visible compared with that in conventional banks. The governance structure employed by Islamic banks is likely to be more complicated than that of conventional (Mollah and Zaman 2015; Elnahass et al. 2020). In both bank types, the board of directors is responsible for the implementation of strategic decisions, protection of the shareholders' interest and maximization of the bank value. However, for Islamic banks, under the constrained banking model and the nature of the products/services offered, the board of directors has additional responsibilities related to the establishment of the appropriate Shariah governance framework besides the development of relevant policies to ensure that all activities are conducted in compliance with the Shariah law (Quttainah et al. 2013). Furthermore, for Islamic banks, additional agency costs are likely to be associated with the Islamic banking model. This is due to a peculiar institutional environment in Islamic banks including the special bank-depositors' relationship.⁴

With expectations that social norms in these religious organizations dominate, effective scrutiny by large and independent board of directors is necessary. Hence, the size of the boards of directors and independence of board members in Islamic banks can substantially influence their monitoring skills and promote high information

⁴ With the absence of representation on the board of directors for depositors, Islamic bank managers have full control of the investment process of depositors' funds which suggest high agency problems.

quality for earnings. Therefore, our first hypothesis are stated in alternative forms:

 H_1 : A large and independent board of directors increases the information value of banks' reported earnings. This effect is more pronounced in Islamic banks when compared to conventional banks.

2.2. Audit committee and information value of earnings

The presence of an audit committee strengthens governance, promotes conservatism, and reduces opportunistic earnings management (Xie et al. 2003; Bédard et al. 2004; Sharma and Kuang 2014). Prior literature has established that an effective audit committee is a governance device that aids the board of directors in its monitoring role and enhances financial reporting quality (Pomeroy and Thornton, 2008; Beasley et al., 2009; Bin-Ghanem and Ariff, 2016).

A large audit committee can be seen as an indication of the resources and varied expertise available to the committee to effectively monitor financial reporting practices (Baxter and Cotter 2009). Yang and Krishnan (2005) provide evidence of a significant negative association between audit committee size and discretionary accruals for a sample of 250 U.S. publicly traded firms between 1996 and 2000. Kent et al. (2010) examines a sample of Australian companies and find that higher accruals quality is associated with larger audit committees. García et al. (2012) study a sample of Spanish firms between 2003 and 2006 and conclude that the size of the audit committee has a significant negative association with discretionary accruals. Previous studies in this context have assessed the quality of financial reporting from an opportunistic earnings management perspective (Yang and Krishnan, 2005; Baxter and Cotter, 2009; García et al., 2012; Chen and Zhang, 2014), from a financial restatement perspective (Abbott et al., 2004; Srinivasan, 2005; Carcello et al., 2011), and from the aspect of conservative accounting (Krishnan and Visvanathan, 2008; Sultana, 2015).

Accomplishing key functions depends on the independence of audit committees' members (Klein 2002). Abbott et al. (2004) suggest that audit committee directors' independence is associated with effective monitoring for two reasons: (1) the absence of economic or psychological ties to management that might conflict with their job duties; and (2) the reputational capital preservation/development motivates independent directors to serve as active overseers of the financial accounting processes. Klein (2002) finds that large increases in abnormal accruals accompany reductions in audit committee independence. Abbott et al. (2004) find that the independence of the audit committee exhibits a significant and negative association with the occurrence of a financial restatement. Chang and Sun (2009) and Chen and Zhang (2014) provide strong evidence for a significant negative association between audit committee independence and earnings management.

An audit committee that is large and composed of independent directors is expected to enhance the information value of bank earnings. We maintain similar predictions to that in the first hypothesis. That is to say that the role of large and independent audit committees in controlling managerial behavior is expected to be more noticeable in Islamic banks than in conventional banks particularly, under the assumed dominance of religious norms, complex governance structure and high agency costs related to the Islamic banking business model. This leads to our second set of hypothesis stated in alternative forms:

 H_2 : A large and independent audit committee increases the information value of bank earnings. This effect is more pronounced in Islamic banks when compared to conventional banks.

3. Research methodology

3.1 Sample and data distribution

We initially had 486 conventional banks and 145 Islamic banks from 23 countries. Following Beck et al. (2013), Mollah et al. (2017) and Elnahass et al. (2022a,b), three sample criteria were applied: (1) countries having both, Islamic and conventional banking systems; (2) availability of governance data; and (3) availability of at least three consecutive years of bank data. The final sample comprises 723 bank-year observations from 100 listed banks (61 conventional and 39 Islamic) in 16 countries for the period 2007 to 2015. The relevance of the sample period is that the Basel II Capital Adequacy framework (Basel Committee on Banking Supervision, 2006) became mandatory for IBs in 2007 (see Ariss and Sarieddine 2007; Elnahass et al. 2020a). This period also allows an examination of whether bank managers opportunistically deviate from accounting standards and regulations during the 2007-2008 financial crisis (see Li et al. 2022).

Financial data are collected from DataStream, Bloomberg, and Bankscope and country-specific macroeconomic data are obtained from the World Bank's World Development Indicators. Both corporate governance data (i.e. board and audit committee characteristics and others) alongside Shariah governance data are hand-collected from banks' annual reports. Moreover, prior literature has found association between earnings quality and other board characteristics such as qualifications, age and tenure. However, since the corporate governance and Shariah governance data in this study is hand-collected from banks' annual reports, information on such characteristics are not available for most of the sampled banks. The final sample distribution across countries and the two bank types is presented in Table 1.

[Insert Table 1 here]

3.2 Measuring the information value of earnings

The main assumption in this study is that the various internal governance mechanisms employed in banks affect the information value of reported earnings. Unlike most of the previous attempts in the literature that examine earnings quality from an opportunistic earnings management perspective, this empirical study uses three measures of earnings quality that particularly reflect the information enhancing value of reported earnings. These measures are earnings persistence, ability of current earnings to predict future cash flows, and reliability of loan loss provisions.

3.2.1 Earnings persistence and predictability of future cash flows

Earnings quality is examined using two related but distinct accounting-based attributes: earnings persistence, and predictive ability of earnings. These two attributes relate to the time-series properties of earnings. According to the time-series perspective, earnings can be classified into two elements: a permanent element and a transitory (temporary) element (Easton et al., 2000). Permanent earnings are the product of

business transactions that generate earnings that will continue in the future. Transitory earnings represent irregular (non- recurring) items recorded in the income statement (Goncharov, 2005; Pan, 2007).

Earnings persistence refers to the extent to which current earnings prevail in the earnings series (Dechow and Schrand, 2004), thus, it is associated with the continuation of earnings over time. Persistent earnings are considered desirable as they are permanent, less transitory, and therefore more useful for forecasting future earnings (Schipper and Vincent, 2003; Frankel and Litov, 2009; Parte-Esteban and García, 2014; García-Sánchez et al., 2017). Earnings persistence has been used in prior literature as a measure of earnings quality, as it contributes to the value relevance of information and it is considered a good input for equity valuation models (Ali and Zarowin, 1992; Ramakrishnan and Thomas, 1998; Dechow et al., 2010). The persistence of earnings is determined by the firm's fundamental performance and the accounting measurement system used to evaluate the performance. In addition, firms with persistent earnings will have a more sustainable earnings stream, making it a useful input to equity valuation models. Accordingly, more persistent earnings are of higher quality than less persistence is estimated as the coefficient on current period earnings at time t (*EBT*_t) in a regression of future earnings on current earnings. EBT measures a bank's capacity to utilize its assets to generate earnings before contractual obligations, and it is measured as earnings during a period before taxes.

The second accounting-based attribute is earnings predictability, which refers to the ability of past earnings to predict future earnings (Lipe, 1990). Financial statement users rely on accounting information to make decisions regarding their capital investment. The predictive process requires assessing firms' risks and future cash flows. High quality predictive earnings enable users to make more accurate assessments of future performance and cash flows. Goncharov (2005) states that persistence and predictability are desired outcomes of financial reporting from the perspective of valuation. For this reason, several empirical studies have used these aspects as measures of financial reporting quality (Doyle et al., 2007; Gaio, 2010; Parte-Esteban and García, 2014). However, their analysis excludes banks and financial institutions because of their unique regulatory environment.

The use of earnings persistence and predictability as measures of earnings quality has recently been documented in banking literature. Altamuro and Beatty (2010) examined the effect of internal control provisions mandated by the Federal Depository Insurance Corporation Improvement Act (FDICIA) on banks' financial reporting quality. In particular, they compared financial reporting of banks governed by the FDICIA's internal control provisions to that of others. They found that improvements in internal control monitoring increased earnings persistence and predictability of cash flow, indicating an enhanced quality of financial reporting. Kanagaretnam et al. (2014) examined the relationship between legal, extra-legal, and political institutional factors and earnings quality of banks across 48 countries. To measure the information value of bank financial reporting, they used earnings persistence and the ability of current earnings to predict the next period's cash flow. They provided evidence that stronger legal, extra-legal, and political factors are associated with higher levels of earnings persistence and cash flow predictability. Two cross-country studies by García-Sánchez and García-Meca (2017), and García-Sánchez et al. (2017) use earnings persistence and the ability

of earnings to predict future cash flow as measures for earnings quality in a sample of banks from nine countries. García- Sánchez and García-Meca (2017) found that banks' commitment to corporate social responsibility practices enhance the persistence of earnings and the predictability of cash flow. Moreover, García-Sánchez et al. (2017) examined the role of gender diversity on boards and financial expertise on audit committees in enhancing banks earnings quality. Their results illustrate that women and financial expert directors improve earnings persistence and ability to predict future cash flow in banks. Delis et al. (2018) investigated whether regulatory enforcement actions issued to banks for violations of rules and regulations improved earnings quality. They found that both the risk-related and the accounting- related enforcement actions significantly improve cash flow predictability and earnings persistence in a sample of US banks that were subject to enforcement actions between 2000 and 2010.

We developed a corporate governance index in the above model (GOV) to measure the characteristics of the internal governance mechanisms using principal components, board of directors and audit committees. Various corporate governance indices have been previously developed by many researchers and institutions. However, most of these indices relate to developed economies (Gompers, Ishii and Metrick's G-Index, 2003; FTSE-ISS Corporate Governance index, 2005; ISS Governance QuickScore, 2013). These indices cover different aspects of corporate governance, such as board structure, board responsibility, audit committee, shareholder rights and equity structure. However, since the corporate governance data in this study is hand-collected from banks' annual reports, information on some of these aspects are not available for most of the sampled banks. Hence, the index in this study (GOV) offers a generic measure for the effectiveness of board of directors (size and independence) and audit committee (size and independence).

To study the effect of internal corporate governance mechanisms on earnings persistence and predictability, the following regression models are estimated:

$$EBT_{t+1} = \beta_0 + \beta_1 EBT_{it} + \beta_2 GOV_{it} + \beta_3 GOV_{it} * EBT_{it} + \beta_4 SIZE_{it} + \beta_5 AGE_{it} + \beta_6 BIG4_{it} + \beta_7 CEODUAL_{it} + \beta_8 CAP_{it} + \beta_9 GDP + \beta_{10} COUNTRY_GOV + \beta_{11} \sum Tt_{t=2015}^{2007} + \varepsilon_{it}$$
(1)

$$EBTLLP_{t+1} = \beta_0 + \beta_1 EBT_{it} + \beta_2 GOV_{it} + \beta_3 GOV_{it} * EBT_{it} + \beta_4 SIZE_{it} + \beta_5 AGE_{it} + \beta_6 BIG4_{it} + \beta_7 CEODUAL_{it} + \beta_8 CAP_{it} + \beta_9 GDP + \beta_{10} COUNTRY_GOV + \beta_{11} \sum Tt_{t=2015}^{2007} + \varepsilon_{it}$$
(2)

Where:

EBT_{t+1}	earnings before taxes during year $t+1$ deflated by lagged total
	assets
$EBTLLP_{t+1}$	earnings before taxes and loan loss provisions during year <i>t</i> +1
	deflated by lagged total assets
GOV	two (separate) corporate governance indices measured as the
	sum of the characteristics (i.e. size and independence) for the two
	principal components of internal governance system represented
	by: (i) the board of directors; and (ii) audit committee. The size
	is measured as the total absolute number of members and the

	over the total number of members.				
GOV* EBT	interaction variable used to examine the role of internal				
	governance mechanisms in enhancing the persistence and				
	predictability of earnings				
SIZE	bank size, measured as the natural logarithm of the year-end total				
	assets				
AGE	bank age, measured as the natural logarithm of the number of years				
	the bank has operated in the country				
BIG4	an indicator variable for audit quality that takes 1 if the bank's				
	auditor is a Big Four, and 0 otherwise				
CEODUAL	an indicator variable for CEO duality that takes 1 if the CEO is				
	also the chairman of the board, and 0 otherwise				
CAP	capital adequacy measured as Tier 1 capital				
GDP	the country-prevailing GDP annual growth rate				
COUNTRY_GOV	a country governance index measured as the average of six				
	governance measures - control for corruption, government				
	effectiveness, political stability, regularity quality, the rule of				
	law and voice and accountability				
$Tt_{t=2015}^{2007}$	Year Fixed Effects				
ε	error term				

For the GOV variable, we regress each index separately (i.e. for board of directors and audit committee) across all models. In the above models (1) and (2), the coefficient of interest is the coefficient on the interaction variable GOV*EBT, which is expected to have a positive sign, in line with the claim that large and independent board members/audit committee increase the persistence of earnings and their predictability of future cash flows.

Following earlier studies (Gaio, 2010; Altamuro and Beatty, 2010; Wang and Campbell, 2012; García-Sánchez et al., 2017), the empirical models control for bank-specific factors such as bank size (SIZE), bank age (AGE), and capital adequacy level (CAP). Prior literature also finds that financial reporting quality differs depending on some other corporate governance characteristics such as the external audit quality (Abdelsalam et al., 2016) and whether the CEO is also the chairman of the board (Cornett et al., 2009). Accordingly, the models control the quality of the external audit by introducing an indicator variable (BIG4) for banks audited by a Big Four audit firm. An indicator variable (CEODUAL) is also introduced to reflect CEO duality.

Regarding the country-level factors, prior literature has documented that firms operating in countries with well-developed financial markets and strong legal institutions commit to higher quality financial reports (Gaio, 2010). Additionally, it has been documented that strong country governance and legal environment is associated with higher earnings quality (Leuz et al., 2003; Bushman et al., 2004, Elnahass et al. 2022a). Furthermore, Kanagaretnam et al. (2014) provide evidence that stronger political institutional structures are associated with higher earnings quality. Therefore, our empirical models control country-level factors that may explain variations in financial reporting quality. These include the annual growth rate of GDP to account for macroeconomic conditions. Moreover, to capture between-country differences in governance perceptions, we follow Čihák and Hesse (2010) and Trinh et al. (2020) to introduce a country governance

index (COUNTRY_GOV).

3.2.2. Loan loss provisions and future loan charge-offs

Although the main purpose of loan loss provisions is to reflect expected future loan losses, prior studies provide evidence that managers may use loan loss provisions to pursue other objectives. These objectives range from earnings management (Beatty et al., 2002; Agarwal et al., 2007; Elnahass et al. 2014; Abdelsalam et al. 2016), capital management (Ahmed et al., 1999; Anandarajan et al., 2007; Elnahass et al. 2018), to signalling (Kanagaretnam et al., 2005; Leventis et al., 2012). A weaker association between the current period's loan loss provisions and future loan charge-offs implies the existence of managerial discretionary practices. If a large and independent board/audit committee could improve the validity of loan loss provision, then it is expected to observe a larger association between current period's loan loss provisions and next period's loan charge-offs. To test this argument, the following regression model is estimated:

$$CHGOFF_{t+1} = \beta_0 + \beta_1 LLP_{it} + \beta_2 GOV_{it} + \beta_3 GOV_{it} * LLP_{it} + \beta_4 SIZE_{it} + \beta_5 AGE_{it} + \beta_6 BIG4_{it} + \beta_7 CEODUAL_{it} + \beta_8 CAP_{it} + \beta_9 GDP + \beta_{10} COUNTRY_GOV + \beta_{11} \sum Tt_{t=2015}^{2007} + \varepsilon_{it}$$
(3)

Where:

$CHGOFF_{t+1}$	the loan charge offs during year $t+1$ deflated by lagged total assets
LLP	the loan loss provisions during year t deflated by lagged total assets
GOV*LLP	interaction variable used to examine the role of internal governance mechanisms in enhancing the validity of loan loss provision All other variables are defined in Appendix I

The coefficient of interest in model (3) is the coefficient on the interaction variable GOV*LLP, which is expected to have a positive sign in line with the claim that large and independent board/audit committee tend to improve the validity of loan loss provisions in anticipating next period's loan charge-offs.

In this study, the random-effect GLS estimation technique is used, based on the results from the Hausman Test⁵. The use of this method is also justified by the fact that typical corporate governance variables (board of directors and audit committee characteristics) do not vary much over time. Using fixed-effect estimations would result in massive loss of the degrees of freedom (Baltagi, 2005; Mollah and Zaman, 2015).

4. Results

4.1. Descriptive statistics and correlations

Table 2 presents the descriptive statistics for the full sample (in Panel A), the conventional banks (in Panel B), and the Islamic banks (in Panel C). In addition, the two-sample *t*-tests (comparing means for conventional banks and Islamic banks) are reported in the last column.

⁵ Results of the Hausman Test are reported in Tables 5, 6 and 7.

The mean values of the current reported earnings (EBT_t) are 0.019 and 0.014 for conventional and Islamic banks, respectively. This finding indicates that conventional banks report significantly higher earnings relative to Islamic banks (two-sample *t*-test of -3.603). These results are comparable to those of Abdelsalam et al. (2016) and Elnahass et al. (2018, 2022a) who report similar EBT of 0.018 and 0.014 for conventional banks and Islamic banks, respectively. Regarding the loan loss provisions and the loan chargeoffs, conventional banks, and Islamic banks report comparable figures of 0.006 (0.004) and 0.006 (0.005), respectively. With respect to the internal governance variables, results show that for conventional banks (Islamic banks), the mean board of directors' size (BODSIZE) is 9.655 (10.023), board's independence (BODINDEP) is 0.369 (0.372), audit committee size (ACSIZE) is 3.661 (3.553), and audit committee independence (ACINDEP) is 0.544 (0.534), respectively. These means are reasonably close however, according to the mean comparison test, significant differences exist between the two subsamples for BODSIZE only. These results indicate that IBs have a significantly larger board size compared with their conventional counterparts. These findings are also in line with Elnahass et al. (2020a) and highlight the absence of significant differences between conventional banks and Islamic banks with respect to the independence of both board of directors and audit committees alongside the size of the audit committee. This implies that boards and audit committees have common roles in controlling agency problems within conventional and Islamic banks. However, to address more complex agency issues in Islamic banks, Shariah supervisory boards are established.

For bank-specific variables, results also show that, compared with CBs, IBs are younger in age, smaller in size and they have significantly higher capital adequacy than conventional banks. It was also found that, on average, 83%-86% of the banks in the sample are audited by a Big Four audit firm. This high percentage might be due to the complex nature of the banking activities.

[Insert Table 2 here]

Table 3 presents the Pearson correlation matrix for the full sample. Panel (A) presents the correlation coefficients for the variables used in the persistence and cash flow predictability models, while Panel (B) presents the correlation coefficients for the variables used in the third model (i.e. loan loss provision and future loan charge-offs). Panel (A) reveals significant positive correlations between current earnings (*EBT*_t) and both one-period-ahead earnings before taxes and earnings before taxes and loan loss provisions. Panel (B) also reveals significant positive correlation between current period loan loss provisions and one-period-ahead loan charge-offs. The Pearson correlation matrix presented in Table 3 affirms that multicollinearity does not appear to be a statistical problem.

[Insert Table 3 here]

4.2. Earnings persistence and cash flow predictability tests

Table 4 reports the results for the earnings persistence test for the full sample (in Panel A), conventional banks sample (in Panel B), and Islamic banks sample (in Panel C). Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committees. It is expected that large and independent board /audit committee members increase the information value of banks' earnings, through more persistent earnings. This effect is expected to be more noticed for Islamic banks when compared to conventional banks.

In Panel A, the two models separately test for the role of board of directors (BOD) and audit committee (AC) where GOV= BOD includes the sum of size and the independence of the board while GOV= AC represents the sum of the size and the independence of the committee.

Across all models, the coefficient on EBT_t is positively and significantly associated with future EBT_{t+1} . These findings indicate that earnings reported on average, as well as across the two bank types, are significantly and consistently persistent. We find that the coefficient on the interaction variable $GOV*EBT_t$ shows positive and significant coefficients only for models (1), (2), (5), and (6). These results indicate that large and more independent board/audit committee members improves the persistence of earnings in the full sample and the Islamic banks sub-sample only. For the conventional banks model, the coefficient on the interaction terms shows insignificant result in model (3) and a marginal significance in model (4), implying that board of directors as well as audit committee characteristics have no role in enhancing conventional banks earnings persistence.

[Insert Table 4 here]

Table 5 reports the regression results for the cash flow predictability test for the full sample (in Panel A), conventional banks sample (in Panel B), and Islamic banks sample (in Panel C). Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee. The premise is that large and independent board/audit committee members increase the information value of earnings by using quality current earnings to predict future cash flows.

The regression results show that current EBT_t is positively and significantly associated with future cash flows ($EBTLLP_{t+1}$) across all models. These findings suggest that in both conventional banks and Islamic banks as well as on average effect (i.e. for the full sample), current earnings are able to predict future cash flows.

With respect to different bank types, in models (1), (2), (5) and (6) the coefficient on the interaction variable GOV*EBT shows significantly positive association, indicating that large and independent boards of directors and audit committees have an important role in enhancing the ability of current earnings to predict future cash flows only for full sample and in Islamic banks. However, the coefficients on the interaction variable are not significant in models (3) and (4) for conventional banks. These results indicate

that although current earnings in conventional banks can predict future cash flow, internal governance mechanisms have no significant impact on this predictive power.

[Insert Table 5 here]

4.3. Loan loss provisions and future loan charge-offs

The results for the loan loss provisions and future loan charge-offs tests are reported in Table 6. Panel (A) reports the results for the full sample, Panel (B) reports the results for the conventional banks, while the results for the Islamic banks are reported in Panel (C). Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee.

In Panel (A), the coefficient on the current loan loss provisions (LLP) is positive and significant in all models. This indicates that current loan loss provisions reported by our sampled banks, regardless of the bank type, are positively associated with future loan charge-offs. These findings are consistent with prior studies (Altamuro and Beatty, 2010; Kanagaretnam et al., 2014).

More importantly, results show positive and significant coefficients on the interaction variable GOV*LLP on average for the full sample (i.e. Panel A) and for Islamic banks (i.e. Panel C). This finding emphasizes the role of boards of directors and audit committees in monitoring the financial reporting process, ensuring its reliability, and enhancing the validity of loan loss provisions (Pomeroy and Thornton, 2008; Beasley et al., 2009; Elnahass et al. 2022a). However, the results for conventional banks, in Panel (B), do not provide significant evidence support to the initial expectation on the role of boards of directors and audit committees in conventional banks. Although results show that current loan loss provisions in conventional banks are significantly and positively associated with future loan charge-offs, it appears that this association is not affected neither by the board of directors nor by the audit committees' characteristics.

[Insert Table 6 here]

Overall, our main findings results are consistent with predictions and confirm our two studies hypotheses H_1 and H_2 . Findings highlight the importance of board size and independence in enhancing the information value of banks reported earnings through improving the persistence and predictive ability of earnings. Our results are also in line with prior literature claiming that firms having larger boards and audit committees can benefit from the member's knowledge and expertise to enhance the quality of reported earnings (Dalton et al., 1999; Xie et al., 2003; Chang and Sun, 2009; Chen and Zhang, 2014).

The observed significant and differential effects across the two bank types could be attributable to the unique institutional characteristics and complex agency costs, distinguishing Islamic banks from conventional banks, as prior literature supports the view that a strong institutional environment enhances the quality of financial reporting (McGuire et al., 2012; Kanagaretnam et al., 2015; Abdelsalam et al. 2016; Elnahass et al. 2020).

4.4. Additional analyses: the role of Shariah governance

Decisions of the board of directors (audit committee) can depend much on the effectiveness of Shariah compliance for an Islamic bank. The presence of the Shariah supervisory board (SSB), as part of the governance structure in Islamic banks, is a unique characteristic that distinguishes their governance systems from those of conventional banks. SSB represents an additional layer of governance, beside the traditional governance mechanisms. The existence of the SSB is likely to provide additional assurance to shareholders that the social norms of the bank are preserved. At least in principle, having SSB should provide a deterrent against earnings management practices in Islamic banks. Previous studies demonstrate that the SSB, acting as an additional layer of governance, provides moral monitoring over management, which in turn can reduce agency costs (Abdelsalam et al., 2021). Additionally, a study conducted by Quttainah et al. (2013) found that the existence of a large Shariah board can contribute to lowering earnings management. Elnahass et al. (2022a) reported that SSB significantly reduces earnings management.

Therefore, additional tests are carried out to examine the role of SSB in enhancing the information value of earnings in Islamic banks. Specifically, the models test for the size, financial qualification, and multiple directorships of Shariah supervisory board.

Table 7 reports the regression results for the earnings persistence (Panel A), cash flow predictability (Panel B), and the relationship between loan loss provisions and future loan charge-offs (Panel C). We followed Mollah et al. (2017) and Elnahass et al. (2022a) and employed an index for the different characteristics of SSB, measured as the sum of SSB size, its financial qualification, and the multiple memberships held by its members.

The results show that the coefficient on current EBT is positive and significant (0.5012) and (0.5754) in Panels (A) and (B), respectively. These results indicate that current earnings reported by Islamic banks are persistent and can predict future cash flows. In addition, Panel (C) shows that the coefficient on current loan loss provisions (LLP) is positive and significant (0.3896), suggesting that current provisions reported by banks reflect expected future loan losses. These findings provide consistent evidence that Islamic banks exhibit high information value for reported earnings even controlling for Shariah governance.

When examining whether SSB mediates the predicted association (i.e. have a role in enhancing the information value of earnings), our results do not provide significant evidence. The coefficients on the interaction variables SSB*EBT and SSB*LLP are insignificant across all models. The lack of evidence on Shariah advisors' characteristics to enhance the information value of earnings may be attributable to the common advisory role played by Shariah scholars in the financial reporting process. Bank managers are responsible for the preparation of financial statements while SSB sit on board to review and scrutinize bank activities for Shariah complaint operations. Moreover, the complex nature of Islamic bank's activities and financial reporting require special accounting and financial expertise to effectively monitor managerial financial reporting decisions (Elnahass et al., 2022a). These justifications are also supported by Khalaf (2007) and Trinh et al. (2020) who argue that graduates Shariah scholars who lack financial

education may not be sufficiently competent to perform specific duties related to deep review of financial transactions and accounting treatments. Hence, SSB tends not to have an impact on the information value of reported earnings.

[Insert Table 7 here]

4.5. Robustness checks

To assess the credibility of the main findings, additional tests are performed. Firstly, to account for the "unbalanced" nature of the study sample, all tests examining the role of internal governance mechanisms in enhancing the information value of earnings are re- estimated using only those countries with observations on both types of banks (conventional and Islamic). The results from these additional tests are reported in Tables 8, 9, and 10 and they support the main findings of Tables 4, 5, and 6. These results provide strong and consistent evidence for the role of boards of directors and audit committees in promoting high earnings persistence, cash flow predictability, and reliability of loan loss provisions on average and for Islamic banks while conventional banks consistently show insignificant evidence.

[Insert Tables 8, 9, 10 here]

Secondly, additional tests examine whether government ownership in banks influences the information value of earnings. To accomplish this, the main models are extended to control for the effect of government ownership (GOV_OWN), which is measured as the proportion of shares held by the government. Existing literature has documented that government ownership is associated with poor corporate governance (Shleifer, 1998; Abdelsalam et al. 2020, 2021), however, no evidence exists on the impact of government ownership on information value of earnings for our sampled banks. The results reported in Tables 11, 12, and 13 show that the main findings remain the same after controlling for government ownership. The results demonstrate that government ownership does not mediate the relationship between corporate governance mechanisms and the information value of earnings.

[Insert Tables 11, 12, 13 here]

Thirdly, additional tests are performed to address the issue of insignificant variables. These tests examine whether the signs and values of significant variables change when the insignificant variables are dropped from the analysis. The main results presented in Tables 4, 5, and 6 show that some control variables have insignificant values (i.e. AGE, CEODUAL, and GDP). The results from these additional tests are reported in Tables 14, 15, and 16 and show that the results for the main variables (board size, board independence, audit committee size, audit committee independence) remain unchanged after dropping insignificant control variables.

[Insert Tables 14, 15, 16 here]

Finally, the empirical analyses employ a panel data analysis, and our estimations account for the unobservable and constant heterogeneity (i.e. management style, business strategy or other bank-specific

features). However, some independent variables in the model (e.g. board structure, composition and functioning) are determined simultaneously with dependent variables, leading to possible simultaneity bias. To address the potential endogeneity problem between corporate governance variables (Elnahass et al. 2020; Trinh et al. 2020) and information value of earnings, Two-step system generalized method of moments (GMM) is utilized for additional robustness checks. The GMM controls for the unobserved effects by transforming the variables into first differences to eliminate unobserved heterogeneity and omitted variable bias. It allows us to treat all bank characteristics variables as endogenous and orthogonally employs the lag values of endogenous variables as IVs (Mollah and Zaman 2015; Mollah et al. 2017). Macroeconomics control variables are treated as strictly exogenous.

Unreported results for both GMM estimations show that the main findings presented earlier in Tables 4, 5, and 6 remain unchanged, after controlling for dynamic endogeneity and unobserved heterogeneity.

5. Conclusion

We utilized a unique set of international data for countries operating on dual banking systems to examine whether different board and audit committee characteristics can enhance the information value of banks earnings within an international context. A structured investigation for the characteristics (i.e. size and independence) of key governance instruments (like board of directors and audit committees) on promoting earnings informative still lacks particularly when comparing Islamic and conventional banks. This study is among the first attempts to examine financial reporting quality from an information perspective and with respect to internal governance characteristics. The study also bridges the gap in the existing literature by forming comparative assessments among Islamic and conventional banks and addresses the Shariah governance effect.

Based on systematic empirical analyses for 732 bank-year observations for 16 countries operating on a dual banking system, the main findings support our hypotheses and suggest that banks report high quality of information for reported earnings. We find strong evidence for the full sample that large and independent boards of directors/audit committees significantly increase the information value of earnings. However, depending on the banking, our results consistently show that board of directors' as well as audit committee characteristics have no role in enhancing conventional banks' earnings persistence, predicting future cash flow and reliability of loan loss provisions. This is the opposite case for their Islamic counterparts. Moreover, we find that Shariah boards' characteristics (i.e., size, financial expertise, and multiple directorships) have no direct or significant impact on the informativeness of earnings within Islamic banks. This finding suggests that unique institutional factors within Islamic banks are not prevailing in the standard banking governance system like board and audit committees. These findings imply that, unlike conventional banking, the informativeness of earnings (and more generally higher financial quality) in Islamic banking could be determined by the characteristics of internal governance mechanisms. However, findings suggest that characteristics for Shariah supervisory board (SSB) members seem not to significantly affect the information value of earnings.

Our study extends the current literature on governance of different bank types while providing new insights to global banking, policy makers and market participants. Policymakers and regulators can use the evidence presented in this study to establish effective double governance mechanisms for countries operating on a dual banking system to unify/monitor financial reporting practices and mitigate future financial crises. By providing evidence on the relationship between internal governance mechanisms and earnings quality, the results provide useful information to investors and regulators who seek enhanced information on the value of a bank's reported earnings. Investors and other market participants should consider and maybe assess different board/audit committee characteristics, institutional characteristics, and corporate business orientations for their banks (i.e., religiously oriented). They should incorporate these factors into their decision-making, stock market valuations and investment choices.

This study also substantiates the influence of social norms on core economic matters with important ramifications for financial reporting framework and internal governance systems. Our findings also inform future guidance and regulations on financial reporting quality for dual banking systems. We recognize a limitation for our main findings to study other governance characteristics. This was due to data restrictions/availability for corporate and Shariah governance data given the rigorous hand-collection process involved for our international sample. Therefore, we invite future research to specifically evaluate other board/audit committee characteristics such as age, gender and tenure. Future studies can also explore internal governance mechanisms such as the effect of risk committees' effectiveness, blockholding and institutional ownership for both types of banks.

Acknowledgement:

The co-authors thank the editor and the two anonymous referees for their valuable comments and directions which have helped to improve the overall exposition of the study. This research complies with the stated ethical standards and did not receive any specific grant from funding agencies in the public, commercial, or non-for-profit sectors. There are no conflicts of interest to disclose.

References

- Abbott, L., Parker, S., and Peters, G. (2004). Audit committee characteristics and restatements. *Auditing: A Journal of Practice and Theory*, 23(1), 69-87.
- Abdelsalam, O., Dimitropoulos, P., Elnahass, M., and Leventis, S. (2016). Earnings management behaviors under different monitoring mechanisms: The case of Islamic and conventional banks. *Journal of Economic Behavior and Organization*, 132, 155-173.
- Abdelsalam, O., Elnahass, M., Ahmed, H., & Williams, J. (2020). Asset securitizations and bank stability: evidence from different banking systems. Global finance journal. Doi: <u>https://doi.org/10.1016/j.gfj.2020.100551</u>.
- Abdelsalam, O., Elnahass, M., Batten, J. A., & Mollah, S. (2021). New insights into bank asset securitization: The impact of religiosity. Journal of Financial Stability, 54, 100854.
- Abdul Rahman, A. and Bukair, A. (2013). The influence of the Shariah supervision board on corporate social responsibility disclosure by Islamic banks of Gulf Co-Operation Council countries. *Asian Journal of Business and Accounting*, 6(2), 65-104.
- Adams, R. and Mehran, H. (2012). Bank board structure and performance: Evidence for large bank holding companies. *Journal of Financial Intermediation*, 21(2), 243-267.

- Agarwal, S., Chomsisengphet, S., Liu, C., and Rhee, G. (2007). Earnings management behaviors under different economic environments: Evidence from Japanese banks. *International Review of Economics and Finance*, 16(3), 429-443.
- Ahmed, A., Tekeda, C., and Thomas, S. (1999). Bank loan loss provisions: A Reexamination of capital management, earnings management and signalling effects. *Journal of Accounting and Economics*, 28(1), 1-25.
- Akerlof, G. A. (1980). A theory of social custom, of which unemployment may be one consequence. *The Quarterly Journal of Economics*, 94(4), 749-775.
- Alharbi, R., Elnahass, M., & McLaren, J. (2022). Women directors and market valuation: What are the "Wonder Woman" attributes in banking?. Journal of International Financial Markets, Institutions and Money, 80, 101611.
- Ali, A. and Zarowin, P. (1992). The role of earnings levels in annual earnings-returns studies. *Journal of Accounting Research*, 30(2), 286-296.
- Almutairi, A. R., & Quttainah, M. A. (2017). Corporate governance: evidence from Islamic banks. *Social Responsibility Journal*, 13(3), 601-624.
- Altamuro, J. and Beatty, A. (2010). How does internal control regulation affect financial reporting? *Journal of Accounting and Economics*, 49(1-2), 58-74.
- Anandarajan, A., Hasan, I., and McCarthy, C. (2007). Use of loan loss provisions for capital, earnings management and signalling by Australian banks. *Accounting and Finance*, 47(3), 357-379.
- Ariss, R. T., & Sarieddine, Y. (2007). Challenges in implementing capital adequacy guidelines to Islamic banks, *Journal* of Banking Regulation, 9 (1), 46-59.
- Arnold, B. and De Lange, P. (2004). Enron: An examination of agency problems. *Critical Perspectives on Accounting*, 15(6), 751-765.
- Arnold, B., & De Lange, P. (2004). Enron: an examination of agency problems. Critical Perspectives on Accounting, 15(6), 751-765.

Baltagi, B. (2005). Econometric analysis of panel data. John Wiley and Sons, Chichester.

- Basel Committee on Banking Supervision (BCBS). (2006). *Enhancing corporate governance for banking organisations*. Basel, Switzerland.
- Baxter, P. and Cotter, J. (2009). Audit committees and earnings quality. Accounting and Finance, 49(2), 267-290.
- Beasley, M. (1996). An empirical analysis of the relation between the board of director composition and financial statement fraud. *The Accounting Review*, 71(4), 443-465.
- Beasley, M., Carcello, J., Hermanson, D., and Neal, T. (2009). The audit committee oversight process. *Contemporary Accounting Research*, 26(1), 65-122.
- Beatty, A., Ke, B., and Petroni, K. (2002). Earnings management to avoid earnings declines across publicly and privately held banks. *The Accounting Review*, 77(3), 547-570.
- Beck, T., Demirgüç-Kunt, A., and Merrouche, O. (2013). Islamic vs. conventional banking: Business model, efficiency and stability. *Journal of Banking and Finance*, 37(2), 433-447.
- Bédard, J., Chtourou, S. M., & Courteau, L. (2004). The effect of audit committee expertise, independence, and activity on aggressive earnings management. *Auditing: A Journal of Practice & Theory, 23*(2), 13-35.
- Berger, A., Kick, T. and Schaeck, K. (2014). Executive board composition and bank risk taking. Journal of Corporate Finance, 28, 48-65.
- Beyer, A., Cohen, D., Lys, T., and Walther, B. (2010). The financial reporting environment: Review of the recent literature. *Journal of Accounting and Economics*, 50(2-3), 296-343.
- Bhattacharya, U., Daouk, H., and Welker, M. (2003). The world price of earnings opacity. *The Accounting Review*, 78(3), 641-678.
- Biddle, G. and Hilary, G. (2006). Accounting quality and firm-level capital investment. *The Accounting Review*, 81(5), 963-982.
- Bin-Ghanem, H. and Ariff, A.M. (2016). The effect of board of directors and audit committee effectiveness on internet financial reporting: Evidence from gulf co-operation council countries. *Journal of Accounting in Emerging Economies*, 6(4), 429-448.
- Bitar, M., Pukthuanthong, K., & Walker, T. (2019). Efficiency in Islamic vs. conventional banking: The role of capital and liquidity. Global Finance Journal, 100487. DOI: <u>https://doi.org/10.1016/j.gfj.2019.100487</u>.
- Brennan, N. (2006). Board of directors and firm performance: Is there an expectations gap? *Corporate Governance: An International Review*, 14(6), 577-593.
- Bushman, R. and Smith, A. (2001). Financial accounting information and corporate governance. *Journal of Accounting and Economics*, 32(1-3), 237-333.
- Bushman, R., Piotroski, J., and Smith, J. (2004). What determines corporate transparency? *Journal of Accounting Research*, 42, 207-252.

- Carcello, J., Neal, T., Palmrose, Z., and Scholz, S. (2011). CEO involvement in selecting board members, audit committee effectiveness, and restatements. *Contemporary Accounting Research*, 28(2), 396-430.
- Cascino, S., Clatworthy, M., García Osma, B., Gassen, J., Imam, S. and Jeanjean, T. (2014). Who uses financial reports and for what purpose? Evidence from capital providers. *Accounting in Europe*, 11(2), 185-209.
- Chang, J. C., & Sun, H. L. (2009). Crossed-listed foreign firms' earnings informativeness, earnings management and disclosures of corporate governance information under SOX. *The International Journal of Accounting*, 44(1), 1-32.
- Chen, C., Collins, D., Kravet, T., and Mergenthaler, R. (2018). Financial statement comparability and the efficiency of acquisition decisions. *Contemporary Accounting Research*, 35(1), 164-202.
- Chen, C., Lu, H., and Sougiannis, T. (2012). The agency problem, corporate governance, and the asymmetrical behavior of selling, general, and administrative costs. *Contemporary Accounting Research*, 29(1), 252-282.
- Chen, J. and Zhang, H. (2014). The impact of the corporate governance code on earnings management Evidence from Chinese listed companies. *European Financial Management*, 20(3), 596-632.
- Chiu, P., Teoh, S., and Tian, F. (2012). Board Interlocks and Earnings Management Contagion. *The Accounting Review*, 88(3), 915-944.
- Čihák, M. and Hesse, H. (2010). Islamic banks and financial stability: An empirical analysis. *Journal of Financial Services Research*, 38(2-3), 95-113.
- Cornett, M. M., McNutt, J. J., & Tehranian, H. (2009). Corporate governance and earnings management at large US bank holding companies. *Journal of Corporate Finance*, 15(4), 412-430.
- Dalton, D., Johnson, J., and Ellstrand, A. (1999). Number of directors and financial performance: A meta-analysis. *Academy of Management Journal*, 42(6), 674-686.
- Davidson, R., Goodwin-Stewart, J., and Kent, P. (2005). Internal governance structures and earnings management. *Accounting and Finance*, 45(2), 241-267.
- De Andres, P. and Vallelado, E. (2008). Corporate governance in banking: The role of the board of directors. Journal of Banking and Finance, 32(12), 2570-2580.
- Dechow, P. and Schrand, C. (2004). Earnings quality. U.S.A.: The Research Foundation of CFA Institute.
- Dechow, P., Ge, W., and Schrand, C. (2010). Understanding earnings quality: A review of the proxies, their determinants and their consequences. *Journal of Accounting and Economics*, 50(2-3), 344-401.
- Delis, M., Hasan, I., Iosifidi, M., and Li, L. (2018). Accounting quality in banking: The role of regulatory interventions. *Journal of Banking and Finance*, 97, 297-317.
- Dimitropoulos, P. E., & Asteriou, D. (2010). The effect of board composition on the informativeness and quality of annual earnings: Empirical evidence from Greece. *Research in International Business and Finance*, 24(2), 190-205.
- Dong, Y., Girardone, C. and Kuo, J. M. (2017). Governance, efficiency and risk taking in Chinese banking. The British Accounting Review, 49(2), 211-229.
- Doyle, J., Ge, W., and McVay, S. (2007). Accruals quality and internal control over financial reporting. *The Accounting Review*, 82(5), 1141-1170.
- Easton, P., Shroff, P., and Taylor, G. (2000). Permanent and transitory earnings, accounting recording lag, and the earnings coefficient. *Review of Accounting Studies*, 5(4), 281-300.
- Elamer, A. A., Ntim, C. G., Abdou, H. A., & Pyke, C. (2019). Sharia supervisory boards, governance structures and operational risk disclosures: Evidence from Islamic banks in MENA countries. Global Finance Journal, 100488. DOI: <u>https://doi.org/10.1016/j.gfj.2019.100488.</u>
- Elnahass, M., Izzeldin, M., and Abdelsalam, O. (2014). Loan loss provisions, bank valuations and discretion: A comparative study between conventional and Islamic banks. Journal of Economic Behavior & Organization, 103, S160-S173.
- Elnahass, M., Izzeldin, M. and Steele, G., 2018. Capital and earnings management: evidence from alternative banking business models. *The International Journal of Accounting*, 53(1), .20-32.
- Elnahass, M., Marie, M., & Elgammal, M. (2022b). Terrorist attacks and bank financial stability: evidence from MENA economies. Review of Quantitative Finance and Accounting, 1-45.
- Elnahass, M., Omoteso, K., Salama, A., & Trinh, V. Q. (2020). Differential market valuations of board busyness across alternative banking models. *Review of Quantitative Finance and Accounting*, 55(1), 201-238.
- Elnahass, M., Salama, A., & Yusuf, N. (2022a). Earnings management and internal governance mechanisms: The role of religiosity. *Research in International Business and Finance*, 59, 101565.
- Fama, E. and Jensen, M. (1983). Separation of ownership and control. Journal of Law and Economics, 26(2), 301-325.
- Feltham, G., Indjejikian, R., and Nanda, D. (2006). Dynamic incentives and dual-purpose accounting. *Journal of* Accounting and Economics, 42(3), 417-437.

- Filatotchev, I. and Wright, M. (2011). Agency perspectives on corporate governance of multinational enterprises. *Journal* of Management Studies, 48(2), 471-486.
- Frankel, R. and Litov, L. (2009). Earnings persistence. Journal of Accounting and Economics, 47(1-2), 182-190.
- Gaio, C. (2010). The relative importance of firm and country characteristics for earnings quality around the world. *European Accounting Review*, 19(4), 693-738.
- García, L. S., Barbadillo, E. R., & Pérez, M. O. (2012). Audit committee and internal audit and the quality of earnings: empirical evidence from Spanish companies. *Journal of Management & Governance, 16*(2), 305-331.
- García-Sánchez, I. and García-Meca, E. (2017). CSR engagement and earnings quality in banks. The moderating role of institutional factors. *Corporate Social Responsibility and Environmental Management*, 24(2), 145-158.
- García-Sánchez, I., Matínez-Ferrero, J., and García-Meca, E. (2017). Gender diversity, financial expertise and its effects on accounting quality. *Management Decision*, 55(2), 347-382.
- Goncharov, I. (2005). *Earnings management and its determinants: Closing gaps in empirical accounting research.* Frankfurt: Peter Lang.
- González, J. and García-Meca, E. (2014). Does corporate governance influence earnings management in Latin American markets? *Journal of Business Ethics*, 121(3), 419-440.
- Hillman, A. J., & Dalziel, T. (2003). Boards of directors and firm performance: Integrating agency and resource dependence perspectives. *Academy of Management Review*, 28(3), 383-396.
- Iatridis, G., & Kadorinis, G. (2009). Earnings management and firms financial motives: A financial investigation of UK listed firms. International Review of Financial Analyses, 18(4), 164-173.
- International Accounting Standards Board (IASB). (2018). *Conceptual Framework for Financial Reporting*. Available at: <u>https://www.iasplus.com/en-gb/news/2018/03/cf</u>, Access date: 30 April 2020.
- Jensen, M. C., & Meckling, W. H. (1976). Theory of the firm: Managerial behavior, agency costs and ownership structure. *Journal of Financial Economics*, 3(4), 305-360.
- Jiraporn, P., Miller, G. A., Yoon, S. S., & Kim, Y. S. (2008). Is earnings management opportunistic or beneficial? An agency theory perspective. *International Review of Financial Analysis*, 17(3), 622-634.
- John, K., De Masi, S., and Paci, A. (2016). Corporate governance in banks. *Corporate Governance: An International Review*, 24(3), 303-321.
- Joshi, P. and Wakil, A. (2004). A study of the audit committees' functioning in Bahrain: Empirical findings. *Managerial Auditing Journal*, 19(7), 832-858.
- Kanagaretnam, K., Lim, C., and Lobo, G. (2014). Effects of international institutional factors on earnings quality of banks. *Journal of Banking and Finance*, 39, 87-106.
- Kanagaretnam, K., Lobo, G. J., & Wang, C. (2015). Religiosity and earnings management: International evidence from the banking industry. *Journal of Business Ethics*, 132(2), 277-296.
- Kanagaretnam, K., Lobo, G., and Yang, D. (2005). Determinants of signalling by banks through loan loss provisions. *Journal of Business Research*, 58(3), 312-320.
- Kang, S. and Kim, Y. (2012). Effect of corporate governance on real activity-based earnings management: Evidence from Korea. *Journal of Business Economics and Management*, 13(1), 29-52.
- Kent, P., Routledge, J., & Stewart, J. (2010). Innate and discretionary accruals quality and corporate governance. *Accounting & Finance, 50*(1), 171-195.
- Khalaf, R. (2007). Islamic banking held back by lack of scholars. *Financial Times*, November 19. Available at: <u>https://www.ft.com/content/75249f32-95df-11dc-b7ec-0000779fd2ac</u>, Access data: 6 May 2020.
- Klein, A. (2002). Audit committee, board of director characteristics, and earnings management. *Journal of Accounting and Economics*, 33(3), 375-400.
- Kohlberg, L. (1984). Philosophy of moral development. Harper and Row, New York.
- Krishnan, G. and Visvanathan, G. (2008). Does the SOX definition of an accounting expert matter? The association between audit committee directors' accounting expertise and accounting conservatism. *Contemporary Accounting Research*, 25(3), 827-858.
- Laeven, L. (2013). Corporate governance: What's special about banks? *Annual Review of Financial Economics*, 5(1), 63-92.
- Lassoued, N., Attia, M., & Sassi, H. (2018). Earnings management in Islamic and Conventional banks: Does ownership structure matter? Evidence from the MENA region. *Journal of International Accounting, Auditing and Taxation, 30*, 85-105.
- Leuz, C., Nanda, D., and Wysocki, P. (2003). Earnings management and investor protection: An international comparison. *Journal of Financial Economics*, 69, 505-527.

- Leventis, S. and Dimitropoulos, P. (2012). The role of corporate governance in earnings management: Experience from US banks. *Journal of Applied Accounting Research*, 13(2), 161-177.
- Leventis, S. and Dimitropoulos, P., and Anandarajan, A. (2012). Signalling by banks using loan loss provisions: The case of the European Union. *Journal of Economic Studies*, 39(5), 604-918.
- Levine, R. (2004). *The corporate governance of banks: A concise discussion of concepts and evidence*. Working paper no. 3404, World Bank Research, Washington, DC.
- Li, F. (2008). Annual report readability, current earnings, and earnings persistence. *Journal of Accounting and Economics*, 45(2-3), 221-247.
- Li, T., Trinh, V. Q., & Elnahass, M. (2022). Drivers of Global Banking Stability in Times of Crisis: The Role of Corporate Social Responsibility. *British Journal of Man*agement. DOI: 10.1111/1467-8551.12631
- Lipe, R. (1990). The relation between stock returns and accounting earnings given alternative information. *The Accounting Review*, 65(1), 49-71.
- McGuire, S., Omer, T., & Sharp, N. (2012). The impact of religion on financial reporting irregularities. *The Accounting Review*, 87(2), 645-673.
- Mollah, S. and Zaman, M. (2015). Shari'ah supervision, corporate governance and performance: Conventional vs. Islamic banks. *Journal of Banking and Finance*, 58, 418-435.
- Mollah, S., Hassan, M., Al Farooque, O. and Mobarek, A. (2017). The governance, risk- taking, and performance of Islamic banks. Journal of Financial Services Research, 51(2), 195-219.
- Owen, A. and Temesvary, J. (2018). The performance effects of gender diversity on bank boards. Journal of Banking and Finance, 90, 50-63.
- Pan, M. (2007). Permanent and transitory components of earnings, dividends, and stock prices. *The Quarterly Review of Economics and Finance*, 47(4), 535-549.
- Parte-Esteban, L. and García, C. (2014). The influence of firm characteristics on earnings quality. *International Journal* of Hospitality Management, 42, 50-60.
- Pathan, S. and Faff, R. (2013). Does board structure in banks really affect their performance? Journal of Banking and Finance, 37(5), 1573-1589.
- Pathan, S. and Skully, M. (2010). Endogenously structured boards of directors in banks. *Journal of Banking and Finance*, 34(7), 1590-1606.
- Peasnell, K., Pope, P., and Young, S. (2005). Board monitoring and earnings management: Do outside directors influence abnormal accruals? *Journal of Business Finance and Accounting*, 32(7-8), 1311-1346.
- Pomeroy, B. and Thornton, D. (2008). Meta- analysis and the accounting literature: the case of audit committee independence and financial reporting quality. *European Accounting Review*, 17(2), 305-330.
- Quttainah, M., Song, L., and Wu, Q. (2013). Do Islamic banks employ less earnings management? *Journal of International Financial Management and Accounting*, 24(3), 203-233.
- Ramakrishnan, R. and Thomas, J. (1998). Valuation of permanent, transitory, and price-irrelevant components of reported earnings. *Journal of Accounting, Auditing and Finance*, 13(3), 301-336.
- Ronen, J., & Yaari, V. (2008). Earnings management (Vol. 372). Springer US.
- Salem, R., Usman, M., & Ezeani, E. (2021). Loan loss provisions and audit quality: Evidence from MENA Islamic and Conventional Banks. The Quarterly Review of Economics and Finance, 79, 345-359.
- Schipper, K. and Vincent, L. (2003). Earnings quality. Accounting Horizons, 17, 97-110.
- Sharma, V. and Kuang, C. (2014). Voluntary audit committee characteristics, incentives, and aggressive earnings management: Evidence from New Zealand. *International Journal of Auditing*, 18(1), 76-89.
- Shleifer, A. (1998). State versus private ownership. *Journal of Economic Perspective*, 12(4), 133-150.
- Shleifer, A. and Vishny, R. (1997). A survey of corporate governance. The Journal of Finance, 52(2), 737-783.
- Srinivasan, S. (2005). Consequences of financial reporting failure for outside directors: Evidence from accounting restatements and audit committee members. *Journal of Accounting Research*, 43(2), 291-334.
- Srivastav, A. and Hagendorff, J. (2016). Corporate governance and bank risk-taking. Corporate Governance: An International Review, 24(3), 334-345.
- Sultana, N. (2015). Audit committee characteristics and accounting conservatism. *International Journal of Auditing*, 19(2), 88-102.
- Tahir, M., Ibrahim, S., & Nurullah, M. (2019). Getting compensation right- The choice of performance measures in CEO bonus contracts and earnings management. *The British Accounting Review*, 51(2), 148-169.
- Talavera, O., Yin, S., & Zhang, M. (2018). Age diversity, directors' personal values, and bank performance. *International Review of Financial Analysis*, 55, 60-79.

- Trinh, V. Q., Elnahass, M., Salama, A., & Izzeldin, M. (2020). Board busyness, performance and financial stability: does bank type matter?. The European Journal of Finance, 26(7-8), 774-801.
- Wang, Y. and Campbell, M. (2012). Corporate governance, earnings management, and IFRS: Empirical evidence from Chinese domestically listed companies. *Advances in Accounting, Incorporating Advances in International Accounting,* 28(1), 189-192.
- Watts, R. and Zimmerman, J. (1978). Towards a positive theory of the determination of accounting standards. *The Accounting Review*, 53(1), 112-134.
- Xie, B., Davidson, W., and DaDalt, P. (2003). Earnings management and corporate governance: The role of the board and the audit committee. *Journal of Corporate Finance*, 9(3), 295-316.
- Yang, J. S., & Krishnan, J. (2005). Audit Committees and Quarterly Earnings Management. International Journal of Auditing, 9(3), 201-219.
- Zalata, A. M., Tauringana, V., & Tingbani, I. (2018). Audit committee financial expertise, gender, and earnings management: Does gender of the financial expert matter?. *International review of financial analysis*, 55, 170-183.

Country	Islamic Banks	Bank-Year Observations	Conventional Banks	Bank-Year Observations	Full Sample	Observations
Bahrain	5	39	2	18	7	57
Bangladesh	6	22	6	34	12	56
Egypt	1	6	1	9	2	15
Indonesia	1	6	8	55	9	61
Jordan	2	14	9	76	11	90
Kuwait	5	35	4	33	9	68
Lebanon	0	0	4	32	4	32
Malaysia	1	9	2	18	3	27
Oman	2	6	3	18	5	24
Pakistan	2	18	2	17	4	35
Palestine	2	14	2	16	4	30
Qatar	3	22	5	42	8	64
Saudi Arabia	4	28	1	9	5	37
Funisia	0	0	3	18	3	18
Гurkey	2	17	7	61	9	78
United Arab Emirates	3	18	2	13	5	31
Banks	39		61		100	
Observations		254		469		723

Notes: This table presents the distribution of the final sample across countries and across the two bank types, after applying the previously discussed sample criteria.

Table 2: Descriptive Statistics													
	PANEL A					PAN	EL B			PAN	EL C		Two-sample <i>t</i> -test
		FULL	SAMPLE		CONVENTIONAL BANKS				ISLAMIC BANKS				(Two Talled)
Variables	Obs.	Mean	Std.	Media	Obs.	Mean	Std.	Media	Obs.	Mean	Std.	Median	T-test
				n				n					
EBT	723	0.017	0.021	0.019	466	0.019	0.014	0.020	257	0.014	0.029	0.015	-3.603***
EBT _{t+1}	723	0.016	0.015	0.018	466	0.018	0.013	0.019	257	0.013	0.018	0.014	-5.020***
EBTLLP _{t+1}	721	0.024	0.015	0.023	465	0.026	0.012	0.024	256	0.020	0.019	0.020	-4.674***
CHGOFF _{t+1}	722	0.004	0.009	0.002	465	0.004	0.010	0.002	257	0.005	0.006	0.003	0.766
LLP	723	0.006	0.009	0.005	466	0.006	0.010	0.004	257	0.006	0.007	0.005	0.460
BODSIZE	723	9.786	2.801	10	466	9.655	2.669	10	257	10.023	3.018	9	1.697*
BODINDEP	661	0.370	0.234	0.333	422	0.369	0.228	0.333	239	0.372	0.244	0.333	0.130
ACSIZE	655	3.624	0.958	3	436	3.661	0.966	3	219	3.553	0.939	3	-1.363
ACINDEP	607	0.541	0.329	0.333	419	0.544	0.328	0.571	188	0.534	0.332	0.667	-0.345
SIZE	723	15.672	1.532	15.732	466	15.934	1.515	16.019	257	15.197	1.449	15.345	-6.355***
AGE	723	3.291	0.787	3.497	466	3.546	0.677	3.714	257	2.828	0.762	2.944	-13.046***
BIG4	723	0.851	0.357	1	466	0.863	0.345	1	257	0.829	0.377	1	-1.222
CEODUAL	723	0.100	0.300	0	466	0.120	0.326	0	257	0.062	0.242	0	-2.497**
CAP	714	16.496	13.565	14.060	460	15.136	6.383	13.670	254	18.959	20.861	15.270	3.635***
GDP	723	4.785	3.942	4.790	466	4.930	3.946	4.876	257	4.522	3.928	4.396	
COUNTRY_GOV	723	-0.165	0.478	-0.089	466	-0.166	0.462	-0.093	257	-0.165	0.508	-0.083	

Notes: This table reports the descriptive statistics. The sample period is 2007 to 2015. Panel A presents the results for the full sample including conventional and Islamic banks with 723 bank-year observations. Panel B presents the results for conventional banks sub-sample comprising 466 bank-year observations. Panel C presents the results for Islamic banks sub-sample comprising 257 bank-year observations. The last column also reports the mean differences and two-sample *t*- test (comparison of means for conventional banks sub-samples).

*,**,*** denote significance at the 10%, 5%, and 1% respectively.

A list of the variables, their definitions, and measures is presented in Appendix I.

Table 3: Pearson Correlation Matrix for the Years 2007-2015																		
Panel (A): Pearson correlation matrix for variables used in persistence and cash flow predictability models.																		
Variables	(1)	(2)	(3)	(4)) (5)	(6))	(7)	(8	3)	(9))	(1	0)	(11)		(12)
1. EBT _{t+1}	1																	
2. EBTLLP _{t+1}	0.80	1																
3. EBT _t	0.68	0.71	1															
4. BOD	0.13	0.13	0.12	1														
5. AC	0.01	0.11	0.04	0.2	2 1													
6. SIZE	0.25	0.28	0.24	0.0	4 0	.17	1											
7. AGE	0.21	0.21	0.15	0.1	2 0	.05	0.4	5 1										
8. BIG4	0.09	0.08	0.06	-0.1	-0	.07	0.4	2	0.25	1								
9. CEODUAL	0.10	0.06	0.02	-0.0	1 -0	.06	-0.0)2	0.12	0.	10 1	L						
10. CAP	-0.15	-0.13	-0.06	-0.0	3 -0	.12	-0. 1	19 ·	0.31	0.	12	-0.0)2	1				
11. GDP	0.17	0.17	0.21	0.0	0 0	.03	0.0)1 ·	-0.08	-0.	07	0.0	3	-0.0)7	1		
12. COUNTRY_GOV	-0.00	0.04	0.04	-0.1	-0	.03	0.4	1	0.08	0.	50	-0.1	19	0.0)6	0.12	1	
Panel (B): Pearson co	orrelation m	atrix for va	riables us	ed in l	oan loss p	provisi	ons a	nd futur	e loan	charge	e-offs i	node	l .					
Variables	(1)	(2)	(3	5)	(4)	(5)	(6)		(7)	(8)	(9	9)	((10)	((11)
1. CHGOFF _{t+1}	1																	
2. LLP	0.48	1																
3. BOD	-0.04	-0.06	5 1														L	
4. AC	0.08	0.09	0.	22 1													<u> </u>	
5. SIZE	-0.06	-0.01	0.	04	0.17	1											L	
6. AGE	-0.05	0.01	0.	12	0.05	0.4	15 1	1									L	
7. BIG4	-0.07	-0.04	- 0 .	14	-0.07	0.4	12	0.25	1								L	
8. CEODUAL	-0.12	-0.09	ə -0.	01	-0.06	-0.0)2	0.12		0.10	1						L	
9. CAP	-0.07	-0.05	5 -0.	03	-0.12	-0.2	19	-0.31		0.12	-0.0)2	1				<u> </u>	
10. GDP	-0.12	-0.18	B 0.	00	0.03	0.0	/1	-0.08	-	0.07	0.0	3	-0	.07	1			
11. COUNTRY_GOV	0.03	0.05	-0	19	-0.03	0.4	1	0.08		0.50	-0.1	19	0.	.06	(0.12	1	

Notes: This table presents the Pearson correlation coefficients for the full sample. Panel (A) presents the correlation coefficients for the variables used in persistence and cash flow predictability models. Panel (B) presents the correlation coefficients for the variables used in the loan loss provision and future loan charge-offs model. Coefficients in bold indicate statistical significance at the 5% significance level or more.

Table 4: Regression Result	Table 4: Regression Results for Earnings Persistence										
Variables	(A) Full S	Sample	(B) Convent	tional Banks	(C) Islami	ic Banks					
	(1) GOV = BOD	(2) GOV = AC	(3) GOV = BOD	(4) GOV = AC	(5) GOV = BOD	(6) GOV =					
	Index	Index	Index	Index	Index	AC Index					
Constant	-0.0078	-0.0071	-0.0046	-0.0012	-0.0097	-0.0329					
	(-0.91)	(-1.06)	(-0.60)	(-0.21)	(-0.42)	(-1.52)					
EBT	0.4184***	0.1828***	0.7448***	0.6160***	0.2174**	0.2009**					
	(3.48)	(3.68)	(6.03)	(7.78)	(2.09)	(2.00)					
GOV	0.0003	-0.0108***	0.0002	-0.0054**	0.0004	-0.0022					
	(1.18)	(-4.37)	(0.91)	(-1.97)	(0.90)	(-1.23)					
GOV * EBT	0.5088**	0.5366***	0.0873	0.1977*	1.5516***	1.5632**					
	(2.25)	(5.47)	(1.30)	(1.67)	(2.68)	(2.42)					
SIZE	0.0006	0.0009**	0.0001	0.0006	0.0017	0.0038***					
	(1.24)	(2.09)	(0.42)	(0.17)	(1.32)	(2.72)					
AGE	0.0009	0.0010	-0.0010	-0.0003	0.0021	0.0031					
	(0.88)	(1.31)	(-1.59)	(-0.37)	(0.91)	(1.09)					
BIG4	0.0034	0.0020	0.0054*	0.0050***	0.0003	-0.0027					
	(1.29)	(1.19)	(1.90)	(3.36)	(0.09)	(-0.81)					
CEODUAL	0.0025	0.0022	-0.0004	-0.0012	0.0028	0.0083*					
	(1.06)	(1.20)	(-0.23)	(-0.84)	(0.97)	(1.85)					
САР	-0.0001**	-0.0001***	-0.0003	-0.0009	-0.0001**	-0.0001*					
	(-2.33)	(-3.51)	(-0.35)	(-0.14)	(-2.28)	(-1.89)					
GDP	0.0001	0.0001	0.0003	0.0007	-0.0003	-0.0004					
	(0.56)	(0.74)	(0.29)	(0.60)	(-0.10)	(-1.29)					
COUNTRY_GOV	-0.0032**	-0.0028**	-0.0028***	-0.0027**	-0.0061**	-0.0098**					
	(-2.41)	(-2.00)	(-2.66)	(-2.52)	(-1.98)	(-2.57)					
EBT + GOV*EBT	0.9272***	0.7193***	0.8321***	0.8137***	1.7690***	1.7641***					
	(6.45)	(12.04)	(13.62)	(14.37)	(3.40)	(3.05)					
Year Dummies	YES	YES	YES	YES	YES	YES					
Adjusted R^Z	56.42%	56.19%	67.98%	66.90%	55.53%	54.08%					
Wald Chi2	239.80***	555.74***	315.08***	743.80***	258.69***	462.40***					
Hausman Test	18.79	14.91	12.00	16.70	15.52	17.45					
Observations	654	604	418	387	236	187					
Notes: This table presents the r Islamic banks sample (Panel C)	regression results for th Columns 1, 3, and 5 re	e first model (i.e. ea port the results for l	rnings persistence for	the full sample (Panel while columns 2, 4, an	l A), conventional bank d 6 present the results	s sample (Panel B), and for audit committee, Z-					

Notes: This table presents the regression results for the first model (i.e. earnings persistence for the full sample (Panel A), conventional banks sample (Panel B), and Islamic banks sample (Panel C). Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee. Z-statistics are between parentheses. *,**,*** denote significance at the 10%, 5% and 1% respectively. All variables are defined in Appendix I. **The estimated Equation:** $EBT_{t+1} = \beta_0 + \beta_1 EBT_{it} + \beta_2 GOV_{it} + \beta_3 GOV_{it} * EBT_{it} + \beta_4 SIZE_{it} + \beta_5 AGE_{it} + \beta_6 BIG4_{it} + \beta_7 CEODUAL_{it} + \beta_8 CAP_{it} + \beta_9 GDP + \beta_{10} COUNTRY_GOV + \beta_{11} \sum Tt_{t=2015}^{2007} + \varepsilon_{it}$

Table 5: Regression Results for Cash Flow Predictability										
Variables	(A) Full S	Sample	(B) Conven	tional Banks	(C) Islam	nic Banks				
	(1) GOV = BOD	(2) GOV = AC	(3) GOV = BOD	(4) GOV = AC	(5) GOV = BOD	(6) GOV = AC				
	Index	Index	Index	Index	Index	Index				
Constant	0.0002	0.0021	0.0048	0.0105	-0.0101	-0.0147				
	(0.01)	(0.25)	(0.38)	(0.90)	(-0.63)	(-0.74)				
EBT	0.3061***	0.1988**	0.5260***	0.5364***	0.2958***	0.2303***				
	(3.54)	(2.20)	(7.79)	(7.79)	(7.47)	(5.18)				
GOV	0.0002	-0.0012**	0.0001	-0.0006	0.0008**	0.0001				
	(1.00)	(-2.11)	(0.30)	(-0.02)	(1.97)	(0.12)				
GOV * EBT	0.5243**	0.0432**	0.1152	0.1717	1.1031***	1.3795***				
	(2.16)	(1.97)	(1.07)	(1.59)	(2.73)	(2.70)				
SIZE	0.0009	0.0009	0.0008	0.0005	0.0014	0.0022*				
	(1.15)	(1.50)	(1.34)	(0.92)	(1.38)	(1.69)				
AGE	0.0010	0.0014	-0.0005	-0.0008	0.0034**	0.0056**				
	(0.76)	(1.30)	(-0.53)	(-0.83)	(1.99)	(2.50)				
BIG4	0.0015	0.0017	0.0015	0.0022	0.0046	0.0034				
	(0.63)	(0.93)	(0.38)	(0.50)	(1.35)	(0.89)				
CEODUAL	0.0012	0.0021	-0.0008	-0.0006	0.0063*	0.0020				
	(0.54)	(1.12)	(-0.39)	(-0.36)	(1.81)	(0.40)				
САР	-0.0007	-0.0005	-0.0003*	-0.0002*	-0.0008	-0.0006				
	(-1.16)	(-1.35)	(-1.93)	(-1.73)	(-1.64)	(-1.00)				
GDP	-0.0002	0.0005	-0.0003	-0.0005	0.0006	-0.0002				
	(-0.18)	(0.51)	(-0.21)	(-0.40)	(0.23)	(-0.49)				
COUNTRY_GOV	-0.0032*	-0.0024	-0.0022	-0.0022	-0.0051	-0.0095**				
	(-1.90)	(-1.43)	(-1.16)	(-1.14)	(-1.62)	(-2.52)				
EBT + GOV*EBT	0.8303***	0.2419***	0.6413***	0.7081***	1.3989***	1.6098***				
	(4.63)	(3.50)	(7.11)	(8.81)	(3.58)	(3.24)				
Year Dummies	YES	YES	YES	YES	YES	YES				
Adjusted R^Z	51.66%	51.41%	59.91%	58.67%	56.46%	52.85%				
Wald Chi2	309.97***	365.07***	633.23***	446.41***	215.88***	135.62***				
Hausman Test	13.71	17.26	15.03	16.27	14.86	16.00				
Observations	653	601	417	414	236	187				

Notes: This table presents the regression results for the second model (i.e. cash flow predictability for the full sample (Panel A), conventional banks sample (Panel B), and Islamic banks sample (Panel C). Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee. *Z*-statistics are between parentheses ****** denote significance at the 10% 5% and 1% respectively. **The estimated Equation:**

Z-statistics are between parentheses. *, **, *** denote significance at the 10%, 5% and 1% respectively. The estimated Equation: $EBTLLP_{t+1} = \beta_0 + \beta_1 EBT_{it} + \beta_2 GOV_{it} + \beta_3 GOV_{it} * EBT_{it} + \beta_4 SIZE_{it} + \beta_5 AGE_{it} + \beta_6 BIG4_{it} + \beta_7 CEODUAL_{it} + \beta_8 CAP_{it} + \beta_9 GDP + \beta_{10} COUNTRY_GOV + \beta_{11} \sum Tt_{22015}^{22007} + \varepsilon_{it}$

Table 6: Regression Results Loan Loss Provisions Predictability									
Variables	(A) Full	Sample	(B) Conven	tional Banks	(C) Islan	nic Banks			
	(1) GOV = BOD	(2) GOV = AC	(3) GOV = BOD	(4) GOV = AC	(5) GOV = BOD	(6) GOV = AC Index			
	Index	Index	Index	Index	Index				
Constant	0.0117**	0.0082***	0.0249***	0.0244***	-0.0034	-0.0007			
	(2.56)	(2.86)	(3.15)	(3.20)	(-0.78)	(-0.15)			
LLP	0.2721***	0.3150***	0.1883***	0.1038**	0.4839***	0.5599***			
	(7.30)	(12.26)	(3.91)	(2.12)	(7.38)	(11.01)			
GOV	-0.0131***	0.0003	-0.0004*	-0.0003	0.0002*	0.0007**			
	(-3.75)	(1.29)	(-1.75)	(-0.78)	(1.68)	(2.07)			
GOV * LLP	1.4755***	0.4694**	0.0923	0.3258	0.3932**	0.4274*			
	(4.57)	(2.24)	(0.15)	(0.52)	(2.14)	(1.71)			
SIZE	-0.0003	-0.0002	-0.0006	-0.0007	0.0004	-0.0001			
	(-0.87)	(-1.21)	(-1.24)	(-1.31)	(0.16)	(-0.52)			
AGE	-0.0005	-0.0006*	-0.0007	-0.0010	0.0007	-0.0001			
	(-0.82)	(-1.79)	(-0.60)	(-1.02)	(0.12)	(-0.30)			
BIG4	-0.0016	-0.0012	-0.0036	-0.0036	0.0019***	0.0019**			
	(-1.38)	(-0.67)	(-0.92)	(-0.97)	(2.64)	(2.03)			
CEODUAL	-0.0016	-0.0009	-0.0004	-0.0008	-0.0031**	-0.0015			
	(-1.37)	(-1.12)	(-0.22)	(-0.44)	(-2.04)	(-1.09)			
CAP	-0.0004	-0.0003*	-0.0001*	-0.0002**	-0.0002	-0.0002			
	(-1.44)	(-1.85)	(-1.72)	(-1.99)	(-1.17)	(-1.30)			
GDP	-0.0009	0.0001	-0.0001	-0.0002	0.0001	0.0001			
	(-1.12)	(0.03)	(-0.92)	(-0.13)	(1.49)	(1.09)			
COUNTRY_GOV	0.0020**	0.0003	0.0031**	0.0029**	-0.0008	-0.0003			
	(2.10)	(0.51)	(2.06)	(1.98)	(-1.00)	(-0.32)			
LLP + GOV*LLP	1.7477***	0.7844***	0.2806	0.4296	0.8771***	0.9872***			
	(5.69)	(3.83)	(0.44)	(0.70)	(5.15)	(4.32)			
Year Dummies	YES	YES	YES	YES	YES	YES			
Adjusted R²	27.30%	34.40%	18.23%	13.20%	52.60%	64.78%			
Wald Chi2	171.45***	225.93***	48.22***	33.20**	146.15***	275.21***			
Hausman Test	9.57	12.08	11.39	13.07	10.41	12.49			
Observations	713	601	417	414	236	187			

Notes: This table presents the regression results for the third model (i.e. loan loss provisions reliability), for the full sample (Panel A), conventional banks sample (Panel B), and Islamic banks sample (Panel C). Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee. Z-statistics are between parentheses. *,**,*** denote significance at the 10%, 5% and 1% respectively. **The estimated Equation:** $CHGOFF_{t+1} = \beta_0 + \beta_1 LLP_{it} + \beta_2 GOV_{it} + \beta_3 GOV_{it} * LLP_{it} + \beta_4 SIZE_{it} + \beta_5 AGE_{it} + \beta_6 BIG4_{it} + \beta_7 CEODUAL_{it} + \beta_9 GDP + \beta_1 COUNTRY_GOV + \beta_{11} \sum Tt_{t=2015}^{2007} + \varepsilon_{it}$

Table 7: Additional Tests - Shariah Supervisory Boards within Islamic Banks										
Panel (A) Earn	ings Persistence Model	Panel (B) Cash	Flow Predictability Model	Panel (C) Loan Loss	Provisions Predictability					
Constant	-0.0359**	Constant	-0.0192	Constant	0.0011					
	(-2.26)		(-1.19)		(0.22)					
EBT	0.5012***	EBT	0.5754***	LLP	0.3896***					
	(4.13)		(4.74)		(2.71)					
SSB	-0.0006	SSB	0.0017**	SSB	0.0002					
	(-0.08)		(2.16)		(0.52)					
SSB * EBT	-0.0370	SSB * EBT	-0.0383	SSB * LLP	0.0335					
	(-1.45)		(-1.51)		(1.28)					
SIZE	0.0034***	SIZE	0.0017	SIZE	-0.0001					
	(3.13)		(1.57)		(-0.30)					
AGE	0.0013	AGE	0.0028	AGE	-0.0005					
	(0.80)		(1.64)		(-0.84)					
BIG4	-0.0018	BIG4	0.0042	BIG4	0.0018					
	(-0.49)		(1.16)		(1.55)					
CEODUAL	0.0056	CEODUAL	0.0079**	CEODUAL	-0.0030***					
	(1.60)		(2.27)		(-2.71)					
CAP	-0.0008*	CAP	-0.0003	CAP	-0.0002					
	(-1.72)		(-0.60)		(-1.21)					
GDP	0.0009	GDP	0.0008	GDP	0.0001					
	(0.37)		(0.34)		(1.51)					
COUNTRY_GOV	-0.0076**	COUNTRY_GOV	-0.0042	COUNTRY_GOV	-0.0008					
	(-2.27)		(-1.22)		(-0.77)					
AAOIFI	0.0044	AAOIFI	-0.0003	AAOIFI	0.0007					
	(1.42)		(-0.00)		(0.71)					
Year Dummies	YES	Year Dummies	YES	Year Dummies	YES					
Adjusted R ²	54.32%	Adjusted R ²	57.64%	Adjusted R ²	56.32%					
Wald Chi2	215.09***	Wald Chi2	286.10***	Wald Chi2	215.53***					
Observations	254	Observations	254	Observations	254					

Notes: This table presents the regression results for the additional tests, examining the role of Shariah supervisory board in enhancing the information value of earnings, through enhancing earnings persistence (Panel A), enhancing cash flow predictability (Panel B), and enhancing the predictability of loan loss provisions (Panel C). Z-statistics are between parentheses. *,**,*** denote significance at the 10%, 5% and 1% respectively.

Table 8: Sensitivity Test (1):	۲able 8: Sensitivity Test (1): Earnings Persistence Test – Countries with Observations on Both Bank Types										
Variables	(A) Full	Sample	(B) Conven	tional Banks	(C) Islar	nic Banks					
	(1) GOV = BOD	(2) GOV = AC	(3) GOV = BOD	(4) GOV = AC	(5) GOV = BOD	(6) GOV = AC					
	Index	Index	Index	Index	Index	Index					
Constant	-0.0066	-0.0074	-0.0025	0.0005	-0.0097	-0.0329					
	(-0.83)	(-1.13)	(-0.34)	(0.08)	(-0.42)	(-1.52)					
EBT	0.4094***	0.1853***	0.7196***	0.6009***	0.2174**	0.2009**					
	(3.48)	(3.67)	(5.45)	(7.21)	(2.09)	(2.00)					
GOV	0.0003	-0.0099***	0.0002	-0.0049*	0.0004	-0.0022					
	(1.18)	(-3.89)	(0.96)	(-1.69)	(0.90)	(-1.23)					
GOV * EBT	0.4731**	0.5108***	0.0743	0.1966	1.5516***	1.5632**					
	(2.19)	(5.10)	(1.05)	(1.59)	(2.68)	(2.42)					
SIZE	0.0010*	0.0013***	0.0004	0.0003	0.0017	0.0038***					
	(1.83)	(2.83)	(1.03)	(0.73)	(1.32)	(2.72)					
AGE	0.0008	0.0010	-0.0011*	-0.0004	0.0021	0.0031					
	(0.84)	(1.20)	(-1.66)	(-0.54)	(0.91)	(1.09)					
BIG4	0.0046	0.0031*	0.0065**	0.0057***	0.0003	-0.0027					
	(1.59)	(1.74)	(2.04)	(3.59)	(0.09)	(-0.81)					
CEODUAL	0.0045	0.0050**	0.0008	0.0005	0.0028	0.0083*					
	(1.60)	(2.30)	(0.41)	(0.26)	(0.97)	(1.85)					
САР	-0.0001**	-0.0001***	-0.0002	0.0001	-0.0001**	-0.0001*					
	(-2.28)	(-3.28)	(-0.26)	(0.02)	(-2.28)	(-1.89)					
GDP	0.0009	0.0001	0.0007	0.0001	-0.0003	-0.0004					
	(0.63)	(0.81)	(0.60)	(0.86)	(-0.10)	(-1.29)					
COUNTRY_GOV	-0.0002***	-0.0001***	-0.0001***	-0.0001***	-0.0061**	-0.0098**					
	(-3.01)	(-2.95)	(-2.91)	(-3.00)	(-1.98)	(-2.57)					
Year Dummies	YES	YES	YES	YES	YES	YES					
Adjusted R ²	57.50%	57.18%	68.56%	67.31%	55.53%	54.08%					
Wald Chi2	241.57***	543.81***	769.27***	700.05***	258.69***	462.40***					
Observations	622	576	386	359	236	187					

Notes: This table presents the first sensitivity test under for the first model (i.e. earnings persistence), examining the role of internal governance mechanisms in enhancing earnings persistence for the full sample (Panel A), conventional banks sample (Panel B), and Islamic banks sample (Panel C). The results reflect only countries with observations on both bank types. Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee. Z-statistics are between parentheses. *,**,*** denote significance at the 10%, 5% and 1% respectively.

Table 9: Sensitivity Test (1): Cash Flow Predictability Test – Countries with Observations on Both Bank Types										
Variables	(A) Full	Sample	(B) Conven	tional Banks	(C) Islan	nic Banks				
	(1) GOV = BOD	(2) GOV = AC	(3) GOV = BOD	(4) GOV = AC	(5) GOV = BOD	(6) GOV = AC				
Constant	0.0017				0.0101					
Constant	0.0017	0.0021	0.0052	(1.10)	-0.0101	-0.0147				
	(0.14)	0.27	(0.43)	(1.10)	(-0.03)	(-0.74)				
EBI	0.3128***	0.1985**	0.5003***	0.51/5***	0.2958***	0.2303***				
	(3.59)	(2.15)	(7.82)	(7.74)	(7.47)	(5.18)				
GOV	0.0002	-0.0013**	0.0001	-0.0005	0.0008**	0.0001				
	(1.01)	(-2.13)	(0.42)	(-0.11)	(1.97)	(0.12)				
GOV * EBT	0.4706**	0.0428*	0.0662	0.1321	1.1031***	1.3795***				
	(1.98)	(1.91)	(0.67)	(1.34)	(2.73)	(2.70)				
SIZE	0.0013	0.0013**	0.0013*	0.0009	0.0014	0.0022*				
	(1.63)	(2.17)	(1.87)	(1.48)	(1.38)	(1.69)				
AGE	0.0011	0.0016	-0.0005	-0.0008	0.0034**	0.0056**				
	(0.85)	(1.41)	(-0.52)	(-0.81)	(1.99)	(2.50)				
BIG4	0.0030	0.0033*	0.0036	0.0042	0.0046	0.0034				
	(1.16)	(1.70)	(0.80)	(0.86)	(1.35)	(0.89)				
CEODUAL	0.0029	0.0042**	0.0017	0.0015	0.0063*	0.0020				
	(1.13)	(2.02)	(1.06)	(0.97)	(1.81)	(0.40)				
САР	-0.0006	-0.0004	-0.0002*	-0.0002*	-0.0008	-0.0006				
	(-1.03)	(-1.11)	(-1.95)	(-1.75)	(-1.64)	(-1.00)				
GDP	-0.0003	0.0005	-0.0003	-0.0006	0.0006	-0.0002				
	(-0.20)	(0.46)	(-0.21)	(-0.47)	(0.23)	(-0.49)				
COUNTRY_GOV	-0.0002***	-0.0002***	-0.0002**	-0.0002**	-0.0051	-0.0095**				
	(-3.02)	(-2.87)	(-2.43)	(-2.28)	(-1.62)	(-2.52)				
Year Dummies	YES	YES	YES	YES	YES	YES				
Adjusted R ²	53.99%	53.52%	61.35%	59.80%	56.46%	52.85%				
Wald Chi2	309.45***	368.32***	629.75***	506.83***	215.88***	135.62***				
Observations	621	569	385	382	236	187				

Notes: This table presents the first sensitivity test under the second model (i.e. cash flow predictability), examining the role of internal governance mechanisms in enhancing the predictive power of earnings for the full sample (Panel A), conventional banks sample (Panel B), and Islamic banks sample (Panel C). The results reflect only countries with observations for both bank types. Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee. Z-statistics are between parentheses. *,**,**** denote significance at the 10%, 5% and 1% respectively.

Table 10: Sensitivity Test (1): Loan Loss Provisions Reliability – Countries with Observations on Both Bank Types						
Variables	(A) Full Sa	ample	(B) Conver	ntional Banks	(C) Islam	nic Banks
	(1) GOV = BOD	(2) GOV = AC	(3) GOV = BOD	(4) GOV = AC	(5) GOV = BOD	(6) GOV = AC Index
	Index	Index	Index	Index	Index	
Constant	0.0084*	0.0082***	0.0213***	0.0258***	-0.0034	-0.0007
	(1.88)	(2.67)	(2.58)	(3.08)	(-0.78)	(-0.15)
LLP	0.2732***	0.3124***	0.1807***	0.0974*	0.4839***	0.5599***
	(7.07)	(11.78)	(3.59)	(1.90)	(7.38)	(11.01)
GOV	-0.0130***	0.0003	-0.0004*	-0.0004	0.0002*	0.0007**
	(-3.58)	(1.27)	(-1.71)	(-0.77)	(1.68)	(2.07)
GOV * LLP	1.4649***	0.4728**	0.1158	0.3433	0.3932**	0.4274*
	(4.39)	(2.19)	(0.17)	(0.52)	(2.14)	(1.71)
SIZE	-0.0002	-0.0002	-0.0007	-0.0007	0.0004	-0.0001
	(-0.75)	(-1.14)	(-1.21)	(-1.35)	(0.16)	(-0.52)
AGE	-0.0005	-0.0006*	-0.0006	-0.0010	0.0007	-0.0001
	(-0.78)	(-1.69)	(-0.55)	(-0.91)	(0.12)	(-0.30)
BIG4	-0.0015	-0.0012	-0.0035	-0.0035	0.0019***	0.0019**
	(-1.18)	(-1.56)	(-0.71)	(-0.77)	(2.64)	(2.03)
CEODUAL	-0.0017	-0.0009	-0.0006	-0.0009	-0.0031**	-0.0015
	(-1.26)	(-0.92)	(-0.28)	(-0.39)	(-2.04)	(-1.09)
CAP	-0.0004	-0.0003*	-0.0002*	-0.0002*	-0.0002	-0.0002
	(-1.42)	(-1.78)	(-1.73)	(-1.95)	(-1.17)	(-1.30)
GDP	-0.0009	-0.0005	-0.0001	-0.0002	0.0001	0.0001
	(-1.08)	(-0.01)	(-0.86)	(-0.18)	(1.49)	(1.09)
COUNTRY_GOV	0.0006	0.0003	0.0009*	0.0030*	-0.0008	-0.0003
	(1.58)	(0.47)	(1.68)	(1.84)	(-1.00)	(-0.32)
Year Dummies	YES	YES	YES	YES	YES	YES
Adjusted R ²	26.65%	33.64%	17.39%	12.60%	52.60%	64.78%
Wald Chi2	155.08***	202.24***	39.78***	33.28**	146.15***	275.21***
Observations	671	569	385	382	236	187

Notes: This table presents the first sensitivity test under the third model (i.e. loan loss provisions and future loan charge-offs), examining the role of internal governance mechanisms in enhancing the validity of loan loss provisions for the full sample (Panel A), conventional banks sample (Panel B), and Islamic banks sample (Panel C). The results reflect only countries with observations for both bank types. Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee. Z-statistics are between parentheses. *,**,*** denote significance at the 10%, 5% and 1% respectively.

Table 11: Sensitivity Test (2): Earnings Persistence Test – Government Ownership Effect						
Variables	(A) Full Sample		(B) Conventional Banks		(C) Islamic Banks	
	(1) GOV = BOD	(2) GOV = AC	(3) GOV = BOD	(4) GOV = AC	(5) GOV = BOD	(6) GOV = AC
	Index	Index	Index	Index	Index	Index
Constant	-0.0009	-0.0026	-0.0002	0.0045	0.0007	-0.0166
	(-0.14)	(-0.42)	(-0.03)	(0.81)	(0.03)	(-0.83)
EBT	0.4030***	0.1840***	0.7437***	0.6063***	0.2120**	0.1950**
	(3.25)	(3.70)	(5.35)	(7.67)	(2.07)	(1.97)
GOV	0.0003	-0.0106***	0.0002	-0.0055**	0.0003	-0.0023
	(1.07)	(-4.31)	(0.86)	(-2.02)	(0.71)	(-1.27)
GOV * EBT	0.7344*	0.5283***	-0.0889	0.1956*	1.5494***	1.4757**
	(1.89)	(5.37)	(-0.41)	(1.66)	(2.63)	(2.17)
SIZE	0.0006	0.0009**	0.0002	-0.0003	0.0019	0.0038***
	(1.34)	(2.12)	(0.51)	(-0.07)	(1.42)	(2.74)
AGE	0.0007	0.0011	-0.0010	-0.0007	0.0020	0.0032
	(0.75)	(1.32)	(-1.57)	(-0.09)	(0.88)	(1.10)
BIG4	0.0041	0.0023	0.0057*	0.0051***	0.0008	-0.0020
	(1.50)	(1.37)	(1.88)	(3.36)	(0.24)	(-0.58)
CEODUAL	0.0020	0.0022	-0.0004	-0.0011	0.0026	0.0076*
	(0.91)	(1.18)	(-0.27)	(-0.75)	(0.88)	(1.65)
GOV_OWN	-0.0096	0.0019	0.0054	0.0041	-0.0018	-0.0080
	(-1.13)	(0.47)	(0.93)	(1.48)	(-0.16)	(-0.55)
САР	-0.0002***	-0.0001***	-0.0002	-0.0001	-0.0001**	-0.0001**
	(-2.72)	(-3.31)	(-0.22)	(-0.21)	(-2.30)	(-2.08)
GDP	0.0006	0.0009	0.0003	0.0007	-0.0004	-0.0004
	(0.46)	(0.72)	(0.32)	(0.56)	(-0.12)	(-1.28)
COUNTRY_GOV	-0.0001**	-0.0001**	-0.0001***	-0.0001***	-0.0002**	-0.0003**
	(-2.44)	(-2.27)	(-2.68)	(-2.81)	(-1.98)	(-2.45)
Year Dummies	YES	YES	YES	YES	YES	YES
Adjusted R ²	56.77%	56.29%	68.10%	67.17%	55.41%	53.74%
Wald Chi2	245.49***	554.36***	472.62***	750.94***	245.54***	827.78***
Observations	624	574	408	377	216	167

Notes: This table presents the second sensitivity test (i.e. whether government ownership in banks affects the information value of earnings) under the first model (i.e. earnings persistence), examining the role of internal governance mechanisms in enhancing earnings persistence for the full sample (Panel A), conventional banks sample (Panel B), and Islamic banks sample (Panel C). Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee. Z-statistics are between parentheses. *,**,*** denote significance at the 10%, 5% and 1% respectively.

Table 12: Sensitivity Test (2): Cash Flow Predictability Test – Government Ownership Effect						
Variables	(A) Full Sample		(B) Conventional Banks		(C) Islamic Banks	
	(1) GOV = BOD	(2) GOV = AC	(3) GOV = BOD	(4) GOV = AC	(5) GOV = BOD	(6) GOV = AC
	Index	Index	Index	Index	Index	Index
Constant	0.0066	0.0054	0.0079	0.0133	-0.0014	0.0052
	(0.61)	(0.70)	(0.67)	(1.25)	(-0.09)	(0.27)
EBT	0.2945***	0.1987**	0.5175***	0.5271***	0.2919***	0.2282***
	(3.45)	(2.21)	(7.61)	(7.65)	(7.36)	(5.17)
GOV	0.0002	-0.0013**	0.0001	-0.0007	0.0007*	-0.0001
	(0.78)	(-2.14)	(0.31)	(-0.02)	(1.74)	(-0.11)
GOV * EBT	0.7371**	0.0427*	0.3109	0.4301*	1.1262***	1.1802**
	(2.48)	(1.95)	(1.23)	(1.77)	(2.76)	(2.26)
SIZE	0.0010	0.0010*	0.0009	0.0007	0.0015	0.0018
	(1.39)	(1.70)	(1.47)	(1.10)	(1.43)	(1.30)
AGE	0.0007	0.0014	-0.0005	-0.0008	0.0033*	0.0059***
	(0.56)	(1.26)	(-0.55)	(-0.92)	(1.92)	(2.57)
BIG4	0.0021	0.0021	0.0021	0.0028	0.0053	0.0036
	(0.88)	(1.18)	(0.51)	(0.60)	(1.48)	(0.92)
CEODUAL	0.0007	0.0019	-0.0012	-0.0010	0.0061*	0.0006
	(0.32)	(1.00)	(-0.59)	(-0.56)	(1.75)	(0.12)
GOV_OWN	-0.0150**	-0.0047	-0.0072	-0.0101	0.0006	-0.0303
	(-2.31)	(-0.77)	(-0.94)	(-1.43)	(0.01)	(-1.50)
САР	-0.0008*	-0.0005	-0.0002*	-0.0002*	-0.0008	-0.0009
	(-1.66)	(-1.35)	(-1.87)	(-1.69)	(-1.55)	(-1.43)
GDP	-0.0003	0.0006	-0.0002	-0.0005	0.0005	-0.0010
	(-0.25)	(0.57)	(-0.18)	(-0.38)	(0.22)	(-0.30)
COUNTRY_GOV	-0.0001**	-0.0001*	-0.0010	-0.0009	-0.0002*	-0.0002*
	(-2.13)	(-1.76)	(-1.48)	(-1.39)	(-1.71)	(-1.65)
Year Dummies	YES	YES	YES	YES	YES	YES
Adjusted R ²	53.12%	51.49%	60.38%	59.38%	56.43%	53.03%
Wald Chi2	324.65***	367.64***	675.47***	410.91***	212.74***	137.28***
Observations	623	571	407	404	216	167

Notes: This table presents the second sensitivity test (i.e. whether government ownership in banks affects the information value of earnings) under the second model (i.e. cash flow predictability), examining the role of internal governance mechanisms in enhancing the predictive power of earnings for the full sample (Panel A), conventional banks sample (Panel B), and Islamic banks sample (Panel C). Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee. Z-statistics are between parentheses. *,**,*** denote significance at the 10%, 5% and 1% respectively.

Variables	(A) Full S	ample	(B) Conventional Banks		(C) Islamic Banks	
	(1) GOV = BOD Index	(2) GOV = AC Index	(3) GOV = BOD Index	(4) GOV = AC Index	(5) GOV = BOD Index	(6) GOV = AC Index
Constant	0.0083*	0.0078***	0.0183**	0.0233***	-0.0016	-0.0005
	(1.95)	(2.77)	(2.39)	(3.07)	(-0.42)	(-0.13)
LLP	0.2750***	0.2891***	0.1690***	0.0820*	0.4818***	0.5572***
	(7.38)	(11.10)	(3.45)	(1.65)	(7.23)	(10.73)
GOV	-0.0128***	0.0003	-0.0004*	-0.0004	0.0002	0.0007**
	(-3.68)	(1.31)	(-1.78)	(-0.87)	(1.57)	(2.10)
GOV * LLP	1.4602***	1.1172***	1.3006	1.7558**	0.4480*	0.4234**
	(4.52)	(4.30)	(1.43)	(1.98)	(1.86)	(2.57)
SIZE	-0.0002	-0.0002	-0.0005	-0.0005	0.0004	-0.0010
	(-0.77)	(-0.88)	(-0.96)	(-1.08)	(0.14)	(-0.35)
AGE	-0.0005	-0.0006*	-0.0007	-0.0010	0.0004	-0.0001
	(-0.87)	(-1.68)	(-0.61)	(-0.98)	(0.07)	(-0.30)
BIG4	-0.0016	-0.0012	-0.0035	-0.0036	0.0020***	0.0020**
	(-1.30)	(-1.62)	(-0.86)	(-0.98)	(2.73)	(2.02)
CEODUAL	-0.0017	-0.0011	-0.0007	-0.0009	-0.0032**	-0.0014
	(-1.43)	(-1.42)	(-0.36)	(-0.50)	(-2.05)	(-0.99)
GOV_OWN	-0.0071	-0.0095***	-0.0116*	-0.0135**	-0.0015	0.0005
	(-1.33)	(-4.10)	(-1.87)	(-2.25)	(-0.47)	(0.11)
САР	-0.0004	-0.0003**	-0.0001*	-0.0002*	-0.0002	-0.0002
	(-1.47)	(-2.17)	(-1.66)	(-1.85)	(-1.23)	(-1.04)
GDP	-0.0008	0.0002	-0.0009	0.0006	0.0001	0.0001
	(-1.05)	(0.26)	(-0.73)	(0.01)	(1.45)	(1.10)
COUNTRY_GOV	0.0006*	0.0007	0.0001	0.0034	-0.0003	-0.0002
	(1.73)	(1.23)	(1.03)	(1.33)	(-0.93)	(-0.50)
Year Dummies	YES	YES	YES	YES	YES	YES
Adjusted R²	27.18%	36.47%	18.50%	14.35%	52.75%	64.80%
Wald Chi2	169.84***	247.41***	49.71***	38.56***	145.60***	269.73***
Observations	683	571	407	404	216	167

Notes: This table presents the second sensitivity test (i.e. whether government ownership in banks affects the information value of earnings) under the third model (i.e. loan loss provisions and future loan charge-offs), examining the role of internal governance mechanisms in enhancing the validity of loan loss provisions for the full sample (Panel A), conventional banks sample (Panel B), and Islamic banks sample (Panel C). Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee. Z-statistics are between parentheses.*,*** denote significance at the 10%, 5% and 1% respectively.

Table 14: Sensitivity Test (3): Earnings Persistence Test – Dropping Insignificant Variables							
Variables	(A) Full Sample		(B) Conventional Banks		(C) Islamic Banks		
	(1) GOV = BOD	(2) GOV = AC	(3) GOV = BOD	(4) GOV = AC	(5) GOV = BOD	(6) GOV = AC	
Constant	-0.0069	-0.0058	-0.0036	-0.0006	-0.0052	-0.0262	
	(-0.81)	(-0.88)	(-0.48)	(-0.11)	(-0.24)	(-1.12)	
EBT	0.4226***	0.1894***	0.7425***	0.6046***	0.2135**	0.1813*	
	(3.45)	(3.84)	(15.60)	(7.80)	(2.01)	(1.76)	
GOV	0.0003	-0.0105***	0.0002	-0.0057**	0.0004	-0.0027	
	(1.26)	(-4.30)	(0.71)	(-2.11)	(0.75)	(-1.43)	
GOV * EBT	0.5161**	0.5350***	0.1150	0.2171*	1.6809***	1.6981**	
	(2.17)	(5.48)	(1.58)	(1.87)	(2.85)	(2.53)	
SIZE	0.0007	0.0011***	-0.0009	0.0003	0.0019	0.0039**	
	(1.48)	(2.66)	(-0.34)	(0.10)	(1.34)	(2.34)	
BIG4	0.0041	0.0024	0.0049*	0.0043***	0.0007	-0.0015	
	(1.63)	(1.51)	(1.82)	(3.17)	(0.22)	(-0.40)	
САР	-0.0002**	-0.0001***	-0.0003	-0.0008	-0.0002***	-0.0002**	
	(-2.42)	(-4.08)	(-0.36)	(-0.14)	(-2.67)	(-2.19)	
COUNTRY_GOV	-0.0036**	-0.0030**	-0.0026**	-0.0023**	-0.0065**	-0.0109***	
	(-2.53)	(-2.29)	(-2.52)	(-2.27)	(-2.05)	(-2.65)	
Year Dummies	YES	YES	YES	YES	YES	YES	
Adjusted R ²	55.93%	55.91%	67.77%	66.78%	54.83%	51.17%	
Wald Chi2	198.47***	561.94***	187.93***	745.78***	255.72***	354.34***	
Observations	654	604	418	387	236	187	

Notes: This table presents the third sensitivity test (i.e. whether the signs and values of significant variables change when the insignificant variables are dropped from the analysis) under the first model (i.e. earnings persistence), examining the role of internal governance mechanisms in enhancing earnings persistence for the full sample (Panel A), conventional banks sample (Panel B), and Islamic banks sample (Panel C). Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee.

Z-statistics are between parentheses. *,**,*** denote significance at the 10%, 5% and 1% respectively.

Table 15: Sensitivity Test (3): Cash Flow Predictability- Dropping Insignificant Variables							
Variables	(A) Ful	l Sample	(B) Conventional Banks		(C) Islamic Banks		
	(1) GOV = BOD	(2) GOV = AC	(3) GOV = BOD	(4) GOV = AC	(5) GOV = BOD	(6) GOV = AC	
Constant	-0.0005	0.0021	0.0049	0.0102	0.0004	-0.0054	
	(-0.04)	(0.25)	(0.39)	(0.89)	(0.03)	(-0.27)	
EBT	0.3069***	0.2006**	0.5269***	0.5328***	0.2789***	0.2262***	
	(3.46)	(2.22)	(7.93)	(8.09)	(7.28)	(5.33)	
GOV	0.0003	-0.0012**	0.0008	-0.0002	0.0007*	-0.0004	
	(1.04)	(-2.10)	(0.24)	(-0.05)	(1.72)	(-0.37)	
GOV * EBT	0.5294**	0.0430**	0.1242	0.1825	1.4075***	1.4720***	
	(2.10)	(1.96)	(1.10)	(1.61)	(3.66)	(2.93)	
SIZE	0.0011	0.0012**	0.0007	0.0004	0.0016	0.0026*	
	(1.55)	(2.24)	(1.21)	(0.66)	(1.48)	(1.94)	
BIG4	0.0018	0.0020	0.0013	0.0020	0.0052	0.0044	
	(0.75)	(1.15)	(0.32)	(0.45)	(1.50)	(1.12)	
САР	-0.0008	-0.0006*	-0.0002*	-0.0002*	-0.0001***	-0.0001**	
	(-1.33)	(-1.78)	(-1.91)	(-1.71)	(-2.84)	(-2.02)	
COUNTRY_GOV	-0.0036**	-0.0028*	-0.0020	-0.0021	-0.0060*	-0.0109***	
	(-2.05)	(-1.72)	(-1.12)	(-1.17)	(-1.88)	(-2.79)	
Year Dummies	YES	YES	YES	YES	YES	YES	
Adjusted R ²	51.38%	51.18%	59.79%	58.51%	53.06%	48.94%	
Wald Chi2	277.42***	363.35***	581.82***	388.88***	199.58***	120.89***	
Observations	653	601	417	414	236	187	

Notes: This table presents the third sensitivity test (i.e. whether the signs and values of significant variables change when the insignificant variables are dropped from the analysis) under the second model (i.e. cash flow predictability), examining the role of internal governance mechanisms in enhancing the predictive power of earnings for the full sample (Panel A), conventional banks sample (Panel B), and Islamic banks sample (Panel C). Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee.

Z-statistics are between parentheses. *, **, *** denote significance at the 10%, 5% and 1% respectively.

Table 16: Sensitivity Test (3): Loan Loss Provisions Reliability – Dropping Insignificant Variables						
Variables	(A) Full Sample		(B) Conventional Banks		(C) Islamic Banks	
	(1) GOV = BOD	(2) GOV = AC	(3) GOV = BOD	(4) GOV = AC	(5) GOV = BOD	(6) GOV = AC
Constant	0.0108**	0.0082***	0.0244***	0.0250***	-0.0040	-0.0013
	(2.41)	(2.85)	(3.09)	(3.25)	(-0.95)	(-0.31)
LLP	0.2756***	0.3025***	0.1801***	0.0889*	0.4497***	0.5341***
	(7.41)	(11.82)	(3.77)	(1.83)	(6.03)	(10.80)
GOV	-0.0129***	0.0003	-0.0004*	-0.0004	0.0003**	0.0008**
	(-3.68)	(1.30)	(-1.95)	(-0.85)	(2.21)	(2.46)
GOV * LLP	1.5200***	0.5235**	0.2073	0.5118	0.4728**	0.4624*
	(4.72)	(2.49)	(0.33)	(0.83)	(2.55)	(1.90)
SIZE	-0.0004	-0.0004**	-0.0008*	-0.0009**	0.0008	-0.0007
	(-1.30)	(-1.97)	(-1.68)	(-1.96)	(0.32)	(-0.27)
BIG4	-0.0019	-0.0017	-0.0038	-0.0042	0.0019***	0.0014
	(-0.71)	(-1.29)	(-1.11)	(-1.38)	(2.67)	(1.52)
CAP	-0.0003	-0.0002	-0.0001	-0.0002*	-0.0002*	-0.0001
	(-1.12)	(-1.34)	(-1.61)	(-1.93)	(-1.72)	(-1.01)
COUNTRY_GOV	0.0023**	0.0006	0.0031**	0.0033**	-0.0007	-0.0009
	(2.47)	(1.09)	(2.16)	(2.35)	(-0.90)	(-0.10)
Year Dummies	YES	YES	YES	YES	YES	YES
Adjusted R ²	26.65%	33.46%	17.49%	12.12%	50.44%	63.94%
Wald Chi2	166.38***	207.03***	44.97***	59.78***	111.49***	254.40***
Observations	713	601	417	414	236	187

Notes: This table presents the third sensitivity test (i.e. whether the signs and values of significant variables change when the insignificant variables are dropped from the analysis) under the third model (i.e. loan loss provisions and future loan charge-offs), examining the role of internal governance mechanisms in enhancing the validity of loan loss provisions for the full sample (Panel A), conventional banks sample (Panel B), and Islamic banks sample (Panel C). Columns 1, 3, and 5 report the results for board characteristics, while columns 2, 4, and 6 present the results for audit committee. Z-statistics are between parentheses. *,**,*** denote significance at the 10%, 5% and 1% respectively.

Appendix I: Definition of Variables

Variables	Definitions and Measures
EBT _{t+1}	Earnings before taxes during year $t+1$ deflated by lagged total
	assets.
EBT	Earnings before taxes and loan loss provisions during year <i>t</i> +1
LLP _{t+1}	deflated by lagged total assets.
CHGOFF _{t+1}	Loan charge offs during year $t+1$ deflated by lagged total assets.
LLP	Loan loss provisions during year <i>t</i> deflated by lagged total assets.
GOV	Two (separate) corporate governance indices measured as the sum of the characteristics (i.e. size and independence) for the two principal components of internal governance system represented by: (i) the board of directors; and (ii) audit committee. The size is measured as the total absolute number of members and the independence is measured as the ratio of independent members over the total number of members.
BOD	An index for the structure of the board of directors, measured by
	combining its size (BODSIZE) and independence (BODINDEP).
AC	An index for the structure of the audit committee, measured by
	combining its size (ACSIZE) and independence (ACINDEP).
GOV*EBT	An interaction variable used to examine the role of internal governance
	mechanisms in enhancing the persistence and
	predictability of earnings.
GOV*LLP	An interaction variable used to examine the role of internal governance
	mechanisms in enhancing the validity of loan loss
	provision.
SIZE	Bank size, measured as the natural logarithm of the year-end total
	assets.
AGE	Bank age, measured as the natural logarithm of the number of
	years the bank has operated in the country.
BIG4	An indicator variable for audit quality that takes 1 if the bank's
	auditor is a Big Four, and 0 otherwise.
CEODUAL	An indicator variable for CEO duality that takes 1 if the CEO is
	also the chairman of the board, and 0 otherwise.
САР	Capital adequacy, measured as Tier 1 capital.
GDP	The country-prevailing GDP annual growth rate.
COUNTRY_GOV	A country governance index, measured as the average of six governance measures
	- control for corruption, government effectiveness, political stability, regulatory
	quality, the rule of law, and voice and accountability